

दिनांक /Dated: 25-07-2025





बिड दस्तावेज़ / Bid Document

बिड विवरण/Bid Details				
बिड बंद होने की तारीख/समय /Bid End Date/Time	16-08-2025 14:00:00			
बिड खुलने की तारीख/समय /Bid Opening Date/Time	16-08-2025 14:30:00			
बिड पेशकश वैधता (बंद होने की तारीख से)/Bid Offer Validity (From End Date)	180 (Days)			
मंत्रालय/राज्य का नाम/Ministry/State Name	Ministry Of Electronics And Information Technology			
विभाग का नाम/Department Name	Department Of Electronics And Information Technology			
संगठन का नाम/Organisation Name	Unique Identification Authority Of India (uidai)			
कार्यालय का नाम/Office Name	Hq, New Delhi			
वस्तु श्रेणी /Item Category	Financial Audit Services - Review of Financial Statements, Financial Reporting Framework, Audit report; Audit Firm, CA Firm, CAG Empaneled Audit or CA Firm			
अनुबंध अविध /Contract Period	3 Year(s) 1 Day(s)			
बिडर का न्यूनतम औसत वार्षिक टर्नओवर (3 वर्षों का) /Minimum Average Annual Turnover of the bidder (For 3 Years)	1000 Lakh (s)			
उर्न्हीं/समान सेवा के लिए अपेक्षित विगत अनुभव के वर्ष/Years of Past Experience Required for same/similar service	5 Year (s)			
इसी तरह की सेवाओं का पिछला आवश्यक अनुभव है/Past Experience of Similar Services required	Yes			
वर्षों के अनुभव एवं दर्नओवर से एमएसई को छूट प्राप्त है / MSE Exemption for Years Of Experience and Turnover	Yes Complete			
स्टार्टअप के लिए अनुभव के वर्षों और टर्नओवर से छूट प्रदान की गई है / Startup Exemption for Years Of Experience and Turnover	Yes Complete			
विक्रेता से मांगे गए दस्तावेज़/Document required from seller	Experience Criteria, Bidder Turnover, Certificate (Requested in ATC), Additional Doc 1 (Requested in ATC), Additional Doc 2 (Requested in ATC), Additional Doc 3 (Requested in ATC), Additional Doc 4 (Requested in ATC) *In case any bidder is seeking exemption from Experience / Turnover Criteria, the supporting documents to prove his eligibility for exemption must be uploaded for evaluation by the buyer			

बिड वि	वरण/Bid Details
क्या आप निविदाकारों द्वारा अपलोड किए गए दस्तावेज़ों को निविदा में भाग लेने वाले सभी निविदाकारों को दिखाना चाहते हैं? संदर्भ मेनू है/Do you want to show documents uploaded by bidders to all bidders participated in bid?	No
बिड लगाने की समय-सीमा बढ़ाने के लिए आवश्यक न्यूनतम सहभागी विक्रेताओं की संख्या। / Minimum number of bids required to disable automatic bid extension	3
दिनों की संख्या, जिनके लिए बिड लगाने की समय-सीमा बढ़ाई जाएगी। / Number of days for which Bid would be auto-extended	7
बिड से रिवर्स नीलामी सक्रिय किया/Bid to RA enabled	No
बिड का प्रकार/Type of Bid	Two Packet Bid
तकनीकी मूल्यांकन के दौरान तकनीकी स्पष्टीकरण हेतु अनुमत समय /Time allowed for Technical Clarifications during technical evaluation	2 Days
अनुमानित बिड मूल्य /Estimated Bid Value	30000000
मूल्यांकन पद्धति/Evaluation Method	Total value wise evaluation
मूल्य दर्शाने वाला वित्तीय दस्तावेज ब्रेकअप आवश्यक है / Financial Document Indicating Price Breakup Required	Yes
मध्यस्थता खंड/Arbitration Clause	No
सुलह खंड/Mediation Clause	No

ईएमडी विवरण/EMD Detail

आवश्यकता/Required No	आवश्यकता/Required
----------------------	-------------------

ईपीबीजी विवरण /ePBG Detail

एडवाइजरी बैंक/Advisory Bank	State Bank of India
ईपीबीजी प्रतिशत (%)/ePBG Percentage(%)	3.00
ईपीबीजी की आवश्यक अवधि (माह) /Duration of ePBG required (Months).	38

(a).ईएमडी और संपादन जमानत राशि, जहां यह लागू होती है, लाभार्थी के पक्ष में होनी चाहिए। / EMD & Performance securityshould be in favour of Beneficiary, wherever it is applicable.

लाभार्थी /Beneficiary :

CEO

Unique Identification Authority of India (UIDAI), HQ Ministry of Electronics and Information Technology Government of India (GoI) Bangla Sahib Road, Behind Kali Mandir, Gole Market, New Delhi - 110001 (Uidai)

विभाजन/Splitting

बोली विभाजन लागू नहीं किया गया/ Bid splitting not applied.

एमआईआई अनुपालन/MII Compliance

एमआईआई अनुपालन/MII Compliance	Yes

- 1. If the bidder is a Micro or Small Enterprise as per latest orders issued by Ministry of MSME, the bidder shall be exempted from the eligibility criteria of "Experience Criteria" as defined above subject to meeting of quality and technical specifications. The bidder seeking exemption from Experience Criteria, shall upload the supporting documents to prove his eligibility for exemption.
- 2. If the bidder is a Micro or Small Enterprise (MSE) as per latest orders issued by Ministry of MSME, the bidder shall be exempted from the eligibility criteria of "Bidder Turnover" as defined above subject to meeting of quality and technical specifications. If the bidder itself is MSE OEM of the offered products, it would be exempted from the "OEM Average Turnover" criteria also subject to meeting of quality and technical specifications. The bidder seeking exemption from Turnover, shall upload the supporting documents to prove his eligibility for exemption.
- 3. If the bidder is a DPIIT registered Startup, the bidder shall be exempted from the the eligibility criteria of "Experience Criteria" as defined above subject to their meeting of quality and technical specifications. The bidder seeking exemption from Experience Criteria, shall upload the supporting documents to prove his eligibility for exemption.
- 4. If the bidder is a DPIIT registered Startup, the bidder shall be exempted from the the eligibility criteria of "Bidder Turnover" as defined above subject to their meeting of quality and technical specifications. If the bidder is DPIIT Registered OEM of the offered products, it would be exempted from the "OEM Average Turnover" criteria also subject to meeting of quality and technical specifications. The bidder seeking exemption from Turnover shall upload the supporting documents to prove his eligibility for exemption.
- 5. The minimum average annual financial turnover of the bidder during the last three years, ending on 31st March of the previous financial year, should be as indicated above in the bid document. Documentary evidence in the form of certified Audited Balance Sheets of relevant periods or a certificate from the Chartered Accountant / Cost Accountant indicating the turnover details for the relevant period shall be uploaded with the bid. In case the date of constitution / incorporation of the bidder is less than 3-year-old, the average turnover in respect of the completed financial years after the date of constitution shall be taken into account for this criteria.
- 6. Years of Past Experience required: The bidder must have experience for number of years as indicated above in bid document (ending month of March prior to the bid opening) of providing similar type of services to any Central / State Govt Organization / PSU. Copies of relevant contracts / orders to be uploaded along with bid in support of having provided services during each of the Financial year.
- 7. Estimated Bid Value indicated above is being declared solely for the purpose of guidance on EMD amount and for determining the Eligibility Criteria related to Turn Over, Past Performance and Project / Past Experience etc. This has no relevance or bearing on the price to be quoted by the bidders and is also not going to have any impact on bid participation. Also this is not going to be used as a criteria in determining reasonableness of quoted prices which would be determined by the buyer based on its own assessment of reasonableness and based on competitive prices received in Bid / RA process.
- 8. Past Experience of Similar Services: The bidder must have successfully executed/completed similar Services over the last three years i.e. the current financial year and the last three financial years(ending month of March prior to the bid opening): -
- 1. Three similar completed services costing not less than the amount equal to 40% (forty percent) of the estimated cost; or
- 2. Two similar completed services costing not less than the amount equal to 50% (fifty percent) of the estimated cost: or
- 3. One similar completed service costing not less than the amount equal to 80% (eighty percent) of the estimated cost.

एक्सेल में अपलोड किए जाने की आवश्यकता /Excel Upload Required:

Financial Costing of Engagement of CA Firm or Company - <u>1753428548.xlsx</u>

अतिरिक्त योग्यता /आवश्यक डेटा/Additional Qualification/Data Required

Number of Years of firm/company's existence as per ICAI certificate:10 years

Number of years of experience as on date of which at least XX years should be in internal/external audit of companies, PSUs and centrally funded institutions.:5 years

Number of full-time partners/experienced and qualified professionals in full time employment at senior level with experience in handling similar or relevant projects:5 partner

Number of partners/ qualified professionals in full time employment with DISA/CISA qualification1

Number of XX fulltime CA's required and YY professional audit staff20 Chartered Accountants

Price Breakup Format for the bidders to upload for providing break-up of overall project cost:1753428561.pdf

This Bid is based on Quality & Cost Based Selection (QCBS) . The technical qualification parameters are :-

Parameter Name	Max Marks	Cutoff Marks	Qualification Methodology Document
No. of CAs, No. of Partners, Average Annual Turnover,Experience, Certification,Solvency Certificate, etc.	80	56	<u>View File</u>
Proposed Methodology, Resources experience etc	20	1	<u>View File</u>

Total Minimum Qualifying Marks for Technical Score: 57

QCBS Weightage(Technical:Financial):80:20

Interview Venue: UIDAI Head Office, New Delhi

Presentation Venue: UIDAI Head Office. New Delhi

Pre Bid Detail(s)

मूल्य भिन्नता खंड दस्तावेज़/Pre-Bid Date and Time	प्री-बिड स्थान/Pre-Bid Venue
29-07-2025 14:00:00	UIDAI Head Office, New Delhi

Financial Audit Services - Review Of Financial Statements, Financial Reporting Framework, Audit Report; Audit Firm, CA Firm, CAG Empaneled Audit Or CA Firm (1)

तकनीकी विशिष्टियाँ /Technical Specifications

विवरण/ Specification	मूल्य/ Values
कोर / Core	

विवरण/ Specification	मूल्य/ Values
Scope of Work	Review of Financial Statements , Financial Reporting Framework , Audit report
Type of Financial Audit Partner	Audit Firm , CA Firm , CAG Empaneled Audit or CA Firm
Type of Financial Audit	Internal Audit
Category of Work under Financial Audit	Audit of financial statements , Internal control of financial , Internal Control over Financial Reporting
Type of Industries/Functions	Government IT Industry
Frequency of Progress Report	Monthly
MIS Reporting for Financial Audit support	Yes
Frequency of MIS reporting	Monthly
State	NA
District	NA
एडऑन /Addon(s)	
Post Financial Audit Support	NA

अतिरिक्त विशिष्टि दस्तावेज़ /Additional Specification Documents

परेषिती/रिपोर्टिंग अधिकारी /Consignees/Reporting Officer and Quantity

क्र.सं./S.N o.	परेषिती / रिपोर्टिंग अधिकारी / Consignee Reporting/Officer	पता/Address	संसाधनों की मात्रा / To be set as 1	अतिरिक्त आवश्यकता /Additional Requirement
1	Surjeet	110001,5th Floor, UIDAI Headquarters, Bangla Sahib Road, Behind Kali Mandir, Gole Market, New Delhi	1	N/A

क्रेता द्वारा जोड़ी गई बिड की विशेष शर्तें/Buyer Added Bid Specific Terms and Conditions

1. Generic

OPTION CLAUSE: The buyer can increase or decrease the contract quantity or contract duration up to 25 percent at the time of issue of the contract. However, once the contract is issued, contract quantity or contract duration can only be increased up to 25 percent. Bidders are bound to accept the revised quantity or duration

2. Service & Support

3. Service & Support

Escalation Matrix For Service Support : Bidder/OEM must provide Escalation Matrix of Telephone Numbers for Service Support.

4. Certificates

Bidder's offer is liable to be rejected if they don't upload any of the certificates / documents sought in the Bid document, ATC and Corrigendum if any.

5. Buyer Added Bid Specific ATC

Buyer uploaded ATC document Click here to view the file.

6. Buyer Added Bid Specific SLA

Text Clause(s)

A. All the Financial Audit Service contracts placed through GeM shall be governed by the fol lowing set of Terms and Conditions:

- I. General terms and conditions for Goods and Services.
- II. Service STC contained in this document.
- III. BID / Reverse Auction specific ATC
- B. The above terms and conditions are in reverse order of precedence i.e.ATC shall superse de Service-specific STC which shall supersede the GTC, in case of any conflicting provisions

अस्वीकरण/Disclaimer

The additional terms and conditions have been incorporated by the Buyer after approval of the Competent Authority in Buyer Organization, whereby Buyer organization is solely responsible for the impact of these clauses on the bidding process, its outcome, and consequences thereof including any eccentricity / restriction arising in the bidding process due to these ATCs and due to modification of technical specifications and / or terms and conditions governing the bid. If any clause(s) is / are incorporated by the Buyer regarding following, the bid and resultant contracts shall be treated as null and void and such bids may be cancelled by GeM at any stage of bidding process without any notice:-

- 1. Definition of Class I and Class II suppliers in the bid not in line with the extant Order / Office Memorandum issued by DPIIT in this regard.
- 2. Seeking EMD submission from bidder(s), including via Additional Terms & Conditions, in contravention to exemption provided to such sellers under GeM GTC.
- 3. Publishing Custom / BOQ bids for items for which regular GeM categories are available without any Category item bunched with it.
- 4. Creating BoQ bid for single item.
- 5. Mentioning specific Brand or Make or Model or Manufacturer or Dealer name.
- 6. Mandating submission of documents in physical form as a pre-requisite to qualify bidders.
- 7. Floating / creation of work contracts as Custom Bids in Services.
- 8. Seeking sample with bid or approval of samples during bid evaluation process. (However, in bids for <u>attached categories</u>, trials are allowed as per approved procurement policy of the buyer nodal Ministries)
- 9. Mandating foreign / international certifications even in case of existence of Indian Standards without specifying equivalent Indian Certification / standards.
- 10. Seeking experience from specific organization / department / institute only or from foreign / export experience.
- 11. Creating bid for items from irrelevant categories.

- 12. Incorporating any clause against the MSME policy and Preference to Make in India Policy.
- 13. Reference of conditions published on any external site or reference to external documents/clauses.
- 14. Asking for any Tender fee / Bid Participation fee / Auction fee in case of Bids / Forward Auction, as the case may be.
- 15. Buyer added ATC Clauses which are in contravention of clauses defined by buyer in system generated bid template as indicated above in the Bid Details section, EMD Detail, ePBG Detail and MII and MSE Purchase Preference sections of the bid, unless otherwise allowed by GeM GTC.
- 16. In a category based bid, adding additional items, through buyer added additional scope of work/ additional terms and conditions/or any other document. If buyer needs more items along with the main item, the same must be added through bunching category based items or by bunching custom catalogs or bunching a BoQ with the main category based item, the same must not be done through ATC or Scope of Work.

Further, if any seller has any objection/grievance against these additional clauses or otherwise on any aspect of this bid, they can raise their representation against the same by using the Representation window provided in the bid details field in Seller dashboard after logging in as a seller within 4 days of bid publication on GeM. Buyer is duty bound to reply to all such representations and would not be allowed to open bids if he fails to reply to such representations.

All GeM Sellers / Service Providers are mandated to ensure compliance with all the applicable laws / acts / rules including but not limited to all Labour Laws such as The Minimum Wages Act, 1948, The Payment of Wages Act, 1936, The Payment of Bonus Act, 1965, The Equal Remuneration Act, 1976, The Payment of Gratuity Act, 1972 etc. Any non-compliance will be treated as breach of contract and Buyer may take suitable actions as per GeM Contract.

This Bid is governed by the सामान्य नियम और शर्तै/General Terms and Conditions, conditions stipulated in Bid and Service Level Agreement specific to this Service as provided in the Marketplace. However in case if any condition specified in सामान्य नियम और शर्तै/General Terms and Conditions is contradicted by the conditions stipulated in Service Level Agreement, then it will over ride the conditions in the General Terms and Conditions.

जेम की सामान्य शर्तों के खंड 26 के संदर्भ में भारत के साथ भूमि सीमा साझा करने वाले देश के बिडर से खरीद पर प्रतिबंध के संबंध में भारत के साथ भूमि सीमा साझा करने वाले देश का कोई भी बिडर इस निविदा में बिड देने के लिए तभी पात्र होगा जब वह बिड देने वाला सक्षम प्राधिकारी के पास पंजीकृत हो।बिड में भाग लेते समय बिडर को इसका अनुपालन करना होगा और कोई भी गलत घोषणा किए जाने व इसका अनुपालन न करने पर अनुबंध को तत्काल समाप्त करने और कानून के अनुसार आगे की कानूनी कार्रवाई का आधार होगा।/In terms of GeM GTC clause 26 regarding Restrictions on procurement from a bidder of a country which shares a land border with India, any bidder from a country which shares a land border with India will be eligible to bid in this tender only if the bidder is registered with the Competent Authority. While participating in bid, Bidder has to undertake compliance of this and any false declaration and non-compliance of this would be a ground for immediate termination of the contract and further legal action in accordance with the laws.

---धन्यवाद/Thank You---

GOVERNMENT OF INDIA MINISTRY OF ELECTRONICS & INFORMATION TECHNOLOGY UNIQUE IDENTIFICATION AUTHORITY OF INDIA

NEW DELHI



Request for proposal for

"ENGAGEMENT OF A CA FIRM/COMPANY FOR HANDLING INTERNAL AUDIT FUNCTIONS OF UIDAI"

RFP Number: <u>HQ-18017/1/2025-FIN-HQ</u>

Dated: 25th July, 2025

Disclaimer

- 1. The information contained in this request for proposal document ("RFP") or subsequently provided to the Bidders, whether verbally or in documentary or in any other form by or on behalf of the Purchaser or any of its employees or advisors, is provided to the Bidders on the terms and conditions set out in this RFP and all other terms and conditions subject to which such information is provided.
- 2. This RFP is not an agreement and is neither an offer nor an invitation by the Purchaser to the Bidders or any other person. The purpose of this RFP is to provide interested parties with information that may be useful to them in the formulation of their Proposals. The information is not an exhaustive account of statutory requirements and should not be regarded as a complete or authoritative statement of law.
- 3. While reasonable care has been taken in providing information in this RFP, the Bidders are advised not to rely on this information only but also carry out their independent due diligence and risk assessments before submitting their response to this RFP. Further, the Bidders are advised to conduct their own analysis of the information contained in this RFP, carry out their own investigations about the project, the regulatory regime which applies thereto and all matters pertaining to the Purchaser and to seek their own professional advice on the legal, financial and regulatory consequences of entering into an agreement or arrangement relating to this RFP.
- 4. The information contained in this RFP is subject to updation, expansion, revision and amendment prior to the last day of submission of the bids at the sole discretion of the Purchaser. In case any major revisions to this RFP are made by the Purchaser within seven days preceding the last date of submission of the Proposals, the Purchaser may, at its discretion, provide reasonable additional time to the Bidders to respond to this RFP. Neither the Purchaser nor any of its officers, employees nor any advisors nor consultants undertakes to provide any Bidder with access to any additional information or to update the information in this RFP.
- 5. Purchaser Added Bid Specific Terms and Conditions: The Criteria for evaluation of bid, Scope of work, SLA, other terms and conditions of RFP etc will be applicable as prescribed in RFP document. If any Additional Criteria prescribed by GeM portal is conflicting with RFP documents, then, terms and conditions of RFP shall prevail.

Table of Contents

Disclain	ner	2
1 Sec	ction I – Invitation to Bid	8
1.1	Preamble	8
1.2	Schedule for Invitation to Bid	9
1.2.1	Important Dates	10
1.2.2	Other Details	12
1.3	Check list	13
1.3.1	Check list of documents to be uploaded in the Pre-qualification bid	13
1.3.2	Check List of Documents to be uploaded in Technical Bid	13
1.3.3	Check List of Documents be uploaded in the Commercial Bid	
2 Sec	ction II - Instruction to Bidders	15
2.1	Procedure for Submission of Bids	15
2.2	Cost of Bid Document	17
2.3	Bid Securing Declaration (BSD)	17
2.4	Contents of the Bid Document	18
2.5	Clarification of Bid Document	19
2.6	Amendment to the Bid Document	19
2.7	Language of Bids	19
2.8	Documents Comprising the Bids	19
2.9	Bid Price	20
2.10	Firm Price	20
2.11	Discount	20

2.12	Bidder Qualification	20
2.13	Period of Validity of Bids	21
2.14	Revelation of Price	21
2.15	Terms and Conditions of Bidders	21
2.16	Local Conditions	22
2.17	Headings	22
2.18	Conditions for Pre-Qualification of Bidders	22
2.19	Last Date for Receipt of Bids	25
2.20	Modification and Withdrawal of Bids	25
2.21	Address for Correspondence	255
2.22	Opening of Bids by UIDAI	26
2.23	Clarification	26
2.24	Preliminary Examination	26
2.25	Contacting UIDAI	26
2.26	Post Qualification	26
2.27	Criteria for Evaluation of Bids	27
2.28	UIDAI's Right to Vary Scope of Contract at the time of Award	31
2.29	UIDAI's Right to Accept Any Bid and to Reject Any or All Bids	31
2.30	Notification of Award (NOA)	31
2.31	Signing of Contract	31
2.32	Performance Bank Guarantee (PBG)	32

2.33	Confidentiality of the Document	33
2.34	Rejection Criteria	33
2.35 3 Se	Fraud and Corruptionction III - General Conditions of Contract	
3.1	Definitions	35
3.2	Application	36
3.3	Use of Contract Documents and Information	36
3.4	Delivery and Documents	37
3.5	Period of Contract	37
3.6	Payment schedule	38
3.7	Currency of Payment	39
3.8	Change Orders	39
3.9	Contract Amendments	39
3.10	Assignment	39
3.11	Sub-contracts	40
3.12	Delays in the Vendor's Performance	40
3.13	Liquidated Damages	40
3.14	Termination for Default	41
3.15	Force Majeure	41
3.16	Termination for Insolvency	42
3.17	Termination for Convenience	42

3.18	Dispute Resolution	.43
3.19	Language	. 43
3.20	Applicable Law	. 44
3.21	Notices	. 44
3.22	Deductions	. 44
3.23	Taxes and Duties	. 44
3.24	Continuing Support	. 44
3.25	Conflict of Interest	. 44
3.26	Delivery and Risk Purchase	. 45
3.27	Fall Clause	. 45
3.28	Indemnification and Limitation of Liability	46
	General Indemnity	
	IPR Indemnity	
	Conditions for Indemnity	
	Risk Purchase	
	Limitation of Liability	
	Allocation of liability	
3.29	Other obligations of Vendor	
3.30	Termination for Misrepresentations of facts	. 49
3.31	Good Faith	.50
3.32	Suspension	.50
3.33	Intellectual Property Rights	.50
3.34	Documents Prepared by the Bidder to be the Property of the Purchaser	.51

4 Se	ction IV – Contents of the Bids	52
4.1	Pre-qualification	52
4.1.1	Proposal Submission Form	52
4.1.2	Bidder's Organization and Experience	53
4.2	Technical Bid	55
4.2.1	Bid Particulars	55
4.2.2		
4.2.3	Delivery Schedule	59
4.2.4	Brief profile of manpower initially proposed to be deployed for execution of contract	60
4.3	Commercial Bid	61
4.3.1	Commercial Bid Letter	61
4.3.2	Cost of Services Offered	62
5 Se	ction V – Scope of Work	63
5.1	Introduction	63
5.2	Resource Engagement	64
5.2.1	Description of Personnel	64
5.2.2	Approval of Personnel	65
5.3	Qualification of resources	65
5.4 Ma	anpower service	67
6 SE	CTION VI – Appendix	73
6.1	Appendix A- Contract	73
6.2	Appendix B – Performance Bank Guarantee	75
6.3	Appendix C – Bid Securing Declaration Form	78
6.4	Appendix D - Non-Disclosure Agreement	80
6.5	Appendix E – Frequency of Internal Audits and Travels Entitlements	804

1 Section I - Invitation to Bid

1.1 Preamble

This invitation to bid is for "Engagement of a CA Firm/Company for handling Internal Audit Functions of Unique Identification Authority of India".

- **a)** Bidders are advised to study the Bid document carefully. Submission of bids shall be deemed to have been done after careful study and examination of the bid document with full understanding of its implications.
- **b)** Bid offers prepared in accordance with the procedures enumerated in **Clause 2.1** shall be submitted online (complete in all respect) and must be uploaded on https://gem.gov.in not later than the date and time laid down as per **Clause 1.2.1**. Bidders/Contractors are advised to follow the instructions provided in the 'Instructions to the Contractors/Bidder for the e-submission of the bids online through the GeM portal on https://gem.gov.in
- c) Manual bids shall not be accepted.
- d) Not more than one bid shall be submitted by one bidder or bidders having business relationship. Under no circumstances will father and his son(s) or other close relations who have business relationship with one another (i.e when one or more partner(s)/director(s) are common) be allowed to bid for the same contract as separate competitors. A breach of this condition will render the bids of both parties liable to rejection.
- **e)** Bidder who has downloaded the tender from UIDAI website www.uidai.gov.in and through the GeM portal on https://gem.gov.in shall not tamper/modify the tender form including downloaded price bid template in any manner. In case the same is found to be tampered/ modified in any manner, bid will be completely rejected and Bid Securing Declaration will be executed and Bidder is liable to be banned from doing business with UIDAI.
- **f)** Intending bidders are advised to visit UIDAI website www.uidai.gov.in and GeM portal on https://gem.gov.in at regular intervals or at least 3 days prior to closing date of submission of bid for any corrigendum/addendum/ amendment.
- **g)** The hard copy of original instruments in respect of Bid Securing Declaration, must be delivered to the address as mentioned in the clause 1.2 Schedule for Invitation

to Bid on or before bid opening date/time as mentioned in clause 1.2.1 – Important Dates. The proposal submitted by the Bidder shall be liable to be summarily rejected for non-submission of original instrument in respect of Bid Securing Declaration.

- **h)** Bids will be opened as per date/time as mentioned in clause 1.2.1. After opening of Technical Bid, the results of their qualification as well Financial Bid opening date will be intimated.
- i) All Bids must be accompanied by Bid Securing Declaration
- **j)** The conditions for Pre-Qualification of the bidders as mentioned in clause 2.18 should be fulfilled for consideration of the bid.

1.2 Schedule for Invitation to Bid

a)	Name of the Purchaser	Unique Id (UIDAI)	dentification Authority of India	
		Ministry o	of Electronics & Information	
		Technology, G	ovt. of India (GoI)	
		Bangla Sahib Road		
		Behind Kali Ma	andir	
		Gole Market,		
		New Delhi – 13	10001	

b)	Name of the Contact Person for seeking any clarification	Deputy Director (Finance) Unique Identification Authority of India (UIDAI), Head Office, Ministry of Electronics & Information Technology, Govt. of India (GoI), 5th Floor, Bangla Sahib Road, Behind Kali Mandir, Gole Market, New Delhi – 110001 Queries should be submitted via E-mail to dd2.finance-hq@uidai.net.in
c)	Date till which the response to the bid should be valid	180 days from the date of opening of Technical Bids.

1.2.1 Important Dates

The following table provides information regarding the important dates of the Bid processing:

Activity	Date	
Published Date	25th July, 2025	
	As per GeM Portal	
	Venue: UIDAI Hqrs	
Pre-Bid Conference	5 th Floor, Bangla Sahib Road, Behind Kali Mandir,	
Time, Date, & Venue	Gole Market, New Delhi – 110001	
	Note: A maximum of two representatives per Bidder would be allowed to participate. Bidders will have to intimate the details of the attendees at least 2 days prior to the scheduled date of the Pre-bid conference at the email ID:- dd2.finance-hq@uidai.net.in	
	In case any bidder wishes to attend through video conferencing, the same may be intimated for sharing the link.	

	As per GeM Portal
	All the queries should be received on or before the prescribed date & time, through email only with subject line as follows:
	"Internal Audit Functions Outsourcing Pre-Bid Queries
Submission of Queries/ clarifications, if any	_ <bidder's name="">".</bidder's>
	The queries should be submitted as per the format prescribed in clause 2.5. The Pre-Bid queries to be sent at the email ID:
	dd2.finance-hq@uidai.net.in

	As per GeM portal	
Clarification / corrigendum to be uploaded on the GeM Portal	Note: Purchaser shall not be obligated to respond to any or all the queries. Purchaser may, at its sole discretion, choose to publish responses to the pre-bid queries and/or any corrigendum on the GeM portal on https://gem.gov.in or may send the same through email or any other means.	
Bid Submission Start Date	As per GeM portal	
Bid Submission End Date	As per GeM Portal Proposals and supporting documents shall be signed, scanned and uploaded in the format and mode as provided for in the GeM portal on https://gem.gov.in for this RFP. Documents should be signed by the authorized signatory of the Bidder.	

Opening of Technical Bids	As per GeM Portal	
	Venue: UIDAI HO, New Delhi	
	Note: A maximum of two representatives per bidder would be allowed to participate	
Presentation	As per GeM Portal	
Opening of Commercial	As per GeM Portal	
Bids		

Note: The above dates, time and venue may be altered by the Purchaser at its sole discretion after giving prior notice to the Bidders. Some of the information provided in the above table is further elaborated in the subsequent sections of this RFP and the information provided in the table and subsequent sections of this RFP are to be read in conjunction and are to be interpreted harmoniously.

1.2.2 Other Details

Area	Description
Method of Selection	Quality and Cost Based Selection [QCBS]
Availability of RFP Documents	The Request for Proposal (RFP) would be available on the GeM portal on https://gem.gov.in and also on uidai.gov.in under Tender Section

Area	Description
Bid Securing Declaration	Bid Securing Declaration Form (as per format in Appendix C)
Submission of Bid Securing Declaration Form	Bidders shall submit Bid Securing Declaration form in a sealed envelope titled "Bid Securing Declaration" (BSD) "RFP for engagement of a CA Firm/Company for handling Internal Audit Functions" to the address as mentioned in clause 1.2 (b) before the Bid Submission End date. Copy of Bid Securing Declaration needs to be uploaded on GeM portal under the separate cover as provided above. If the hard copy of Bid Securing Declaration (BSD) in original is not received by the prescribed date and time, the Proposal submitted by the Bidder shall be liable to be summarily rejected.
Language of Bid Submission	Proposals should be submitted in English only.
Currency	Currency in which the Bidders may quote the price and will receive payment is INR only.

1.3 Check list

Please check whether the following have been enclosed in the respective covers, namely,

(i) Pre-Qualification Bid (ii) Technical Bid and (iii) Commercial Bid.

1.3.1 Check List of Documents to Be Uploaded in the Pre- Qualification Bidi.e. Packet - 1 (Pre-Qualification Bid, Technical Bid)

As per details referred in clause 2.18.

Sl.	Pre-Qualification Condition	Whether Enclosed	Page
No.		(Y/N)	No.
(i)	Submission and uploaded copy of Bid Securing Declaration (BSD) pursuant to Clause 2.3		
(ii)	Legal Entity/Registration of Firm		
(iii)	Statutory Compliance Certificates (Self-Certificate for compliance with Tax Authorities)		
(iv)	Bidder's Turnover		
(v)	CAG Registration		
(vi)	Bidder's experience (Projects)		
(vii)	Professional Strength		
(viii)	Registered/Branch Office		
(ix)	Declaration on non-blacklisting		
(x)	Proposal Submission form (clause 4.1.1)		
(xi)	Bidder organization and experience (clause 4.1.2)		
(xii)	Declaration for authorized signatory as		
	mentioned in clause 2.12		

Important Note: Above list should be duly filled and uploaded

1.3.2 Check List of Documents to be Uploaded in Technical Bid i.e. Packet- 1 (Pre-Qualification Bid, Technical Bid)

S.No.	Description	Whether Enclosed (Y/N)	PageNo.
(i)	Bid Particulars (Clause 4.2.1)		
(ii)	Technical Bid letter (Clause 4.2.2)		
(iii)	Delivery Schedule (Clause 4.2.3)		
(iv)	Brief profile of manpower proposed (Clause 4.2.4)		
(v)	Un-priced BoQ (format as mentioned in BoQ_xxxx.xls but without any cost mentioned)		

Important Note: Above list should be duly filled and uploaded

1.3.3 Check List of Documents be uploaded in the Commercial Bid i.e. Packet -2 (Commercial Bid)

S.	Description	Whether	Page
No.		Enclosed	No.
		(Y/N)	
(i)	Commercial bid letter (Clause 4.3.1)		
(ii)	Cost of Services Offered (Clause 4.3.2)		
(iii)	Filled Priced BoQ (excel file)		

Important Note: Above list, duly filled in, should also be uploaded

2 Section II - Instruction to Bidders

2.1 Procedure for Submission of Bids

a. A two packet Bid System will be followed for this RFP which includes Pre-Qualification Bid, Technical Bid and Commercial Bid. Selection will be done as per QCBS method.

Packet-1 (Pre-Qualification and Technical bid) having viz.,

- Part I Pre qualification Bid Submission (Bid Securing Declaration (BSD) + Checklist with Y/N as mentioned in clause 1.3.1 + Documents as per check list)
- **Part II** Technical Bid Submission (Checklist with Y/N as mentioned in clause 1.3.2 + Documents as per check list)
- Sequence of documents to be followed as per checklist and proper index to be provided.

Packet-2 (Commercial Bid) having viz.,

- Financial Bid Submission (Checklist with Y/N as mentioned in 1.3.3 Documents as per check list)
- Schedule of price bid in the form of BoQ format
- Sequence of documents to be followed as per checklists and index to be provided.
- **b.** This RFP process will be administered through the Government e Marketplace ("GeM/Portal") portal https://gem.gov.in. The Bidders are required to submit soft copies of their proposals electronically on the Portal, using valid Digital Signature Certificates.
- c. All the pages of the Bid must be sequentially numbered and must contain the list of contents with page numbers. Any deficiency in the documentation may result in rejection of the Bid. Failure to submit the Bid on time shall cause a proposal to be rejected. The Purchaser will not accept delivery of the Bid by e-mail or any other electronic/non-electronic means other than uploading on Govt. e-marketplace (GeM) portal https://gem.gov.in.

- **d.** The Bidder should take into account any corrigendum to this RFP document that may have been published before submitting their Bids.
- e. The Bidders are requested to go through the RFP carefully to understand the documents required to be submitted and the process to be followed as a part of the Bid. Any deviations may lead to rejection of the Bid. The Bid documents can generally be in the PDF/XLS/RAR formats. The Bid documents may be scanned with 100 dpi with black and white option.
- f. The Bidder should try to submit the Bid well before the last date to avoid any inconvenience at the last moment. The Bid submission date and time will be as mentioned in the clause 1.2 of this RFP. The Bidder will not be allowed to submit the Bid after the Bid submission time. The Bidder has to digitally sign and upload the required Bid documents one by one as indicated in this RFP.
- **g.** A standard BoQ format has been provided with this RFP document to be filled by all the Bidders. The Bidders should necessarily submit their Financial Bids in the format provided and no other format will be acceptable. The BoQ format for this RFP can be downloaded from GeM portal.
- **h.** All the documents being submitted by the Bidders will be encrypted using PKI encryption techniques to ensure the secrecy of data. The data entered cannot be viewed by unauthorized persons until the time of Bid opening. The confidentiality of the Bids is maintained using the secured socket layer 128 bit encryption technology. Data storage encryption for sensitive fields is done.

The uploaded Bid documents become readable only after the Bid opening by authorized Bid openers.

Bidders are advised to study the Bid document carefully. Online Submission of bids shall be deemed to have been done after careful study and examination of the bid document with full understanding of its implications. Bid offers prepared

in accordance with the procedures enumerated in clause 2.1 (a) should be submitted online only through GeM <u>website</u>: https://gem.gov.in not later than the date and time laid down in the Schedule for Invitation to Bid under Clause 1.2.

2.2 Cost of Bid Document

- 1 The RFP is available to be downloaded online, free of cost.
- The Bidder shall bear all costs associated with the preparation and submission of its Bids, including cost of presentation for the purposes of clarification of the Bids, if so desired by UIDAI. UIDAI will, in no case be responsible or liable for those costs, regardless of the conduct or outcome of the bidding process.

2.3 Bid Securing Declaration (BSD)

- **a.** The Bidders shall submit Bid S0065curing Declaration (As per Appendix C) in a sealed envelope.
- **b.** Copy of the Bid Securing Declaration should be uploaded onto the GeM Portal and the hardcopy (signed) of the same shall be submitted as per guidelines mentioned in clause 2.1.
- c. In case the Bid Securing Declaration is not received by the stipulated time then the Purchaser reserves the right to forthwith and summarily reject the Proposal of the Bidder without providing any opportunity for any further correspondence by the Bidder.
- **d.** The Bid Securing Declaration shall be executed :
 - If a Bidder withdraws the Proposal or increases the quoted prices during the period of Bid validity period or extended period, if any.
 - If the Bidder varies or modifies its Proposal in a manner not acceptable to the Purchaser after opening of Bid during the validity period or any extension thereof.
 - In case of a successful Bidder, if the Bidder fails to sign the Agreement in accordance with the terms and conditions (including timelines for execution of the Agreement) of this RFP or fails to furnish the

- Performance Bank Guarantee in accordance with the terms and conditions (including timelines for furnishing PBG) of this RFP.
- During the Bid process, if a Bidder indulges in any act as would jeopardize, influence or unnecessarily delay the process of bid evaluation and finalization.
- **e.** The decision of the Purchaser regarding execution of Bid Securing Declaration shall be final and binding on the Bidders and shall not be called upon in question under any circumstances.

2.4 Contents of the Bid Document

- The Schedule of Requirements of the Licenses and services required, Bid procedures and contract terms are prescribed in the Bid Document. The Bid Document includes:
 - a) Section I Invitation to Bid;
 - b) Section II Instructions to Bidders;
 - c) Section III- General Conditions of Contract;
 - d) Section IV Contents of the Bid
 - i. Pre-qualification
 - ii. Technical Bid
 - iii. Commercial Bid
 - e) Section V- Scope of Work;
 - f) Section VI- Appendices
 - i. Contract Form (Appendix A)
 - ii. Performance Bank Guarantee Form (Appendix B)
 - iii. Bid Securing Declaration Form (Appendix C)
 - iv. Non-Disclosure Agreement Form (Appendix D)
 - v. Frequency of Internal Audits and Travel Entitlements (Appendix E)
- The Bidder is expected to examine complete bid document including all instructions, forms, general terms & conditions, and schedule of requirements in the Bid Document. Failure to furnish any information required by the Bid Document or submission of a bid not substantially responsive to the Bid Document in every respect will be at the Bidder's risk and may result in the rejection of the Bid.

2.5 Clarification of Bid Document

A prospective Bidder requiring any clarification of the Bid Document may notify UIDAI in writing at UIDAI's mail address indicated in clause 1.2. The queries must be submitted over mail in MS - excel format as below:

Name organiza	of tion	the S	r. Io	Clause no.of RFP	Page number of clause	Existing provision in the clause	Clarification sought

2 UIDAI may at its discretion respond to request for clarifications of the Bid Document, received not later than the date prescribed in clause 1.2.1 of this Bid document.

2.6 Amendment to the Bid Document

- At any time prior to the last date and time for receipt of bids, UIDAI may, for any reason, whether at its own initiative or in response to a clarification requested by a prospective Bidder, modify the Bid Document by an addendum/corrigendum.
- The addendum/corrigendum will be notified by UIDAI which will be binding on all the bidders.
- In order to provide prospective Bidders reasonable time to take the addendum/corrigendum into account in preparing their bids, UIDAI may, at its discretion, extend the last date for the submission of Bids.

2.7 Language of Bids

The Bids prepared by the Bidder and all correspondence and documents relating to the bids exchanged by the Bidder and UIDAI, shall be written in English language, provided that any printed literature furnished by the Bidder may be written in another language so long the same is accompanied by an English translation in which case, for purposes of interpretation of the bid, the English translation shall prevail.

2.8 Documents Comprising the Bids

The Bids prepared by the Bidder shall comprise of the following components:

- a) Pre-Qualification Bid
- b) The Technical Bid
- c) The Commercial Bid

2.9 Bid Price

- The Bidder shall indicate in the proforma prescribed at Annexure 4.3.2, the unit cost and total Bid value of the Services, it proposes to provide under the Contract.
- 2 The unit cost quoted in the above mentioned proforma will be used to calculate cost for 'change orders', if any.
- 3 In the absence of any information in the proforma prescribed, as requested above, a bid may be considered incomplete and summarily rejected.
- The Bidder shall prepare the Bid based on details provided in the Bid documents. Bidder shall carry out the study of the engagement of manpower in accordance with the requirements of the Bid document and it shall be the responsibility of the Bidder to fully meet all the requirements of the Bid document.

2.10 Firm Price

- Price quoted must be firm and final and shall not be subject to any upward modifications, on any account whatsoever. The Bidder shall, therefore, indicate the prices in Annexure 4.3.2 enclosed with the Bid. The Bid Prices shall be indicated in Indian Rupees (INR) only.
- The Commercial bid should clearly indicate the price to be charged and should include all taxes, fees and other charges as may be applicable in relation to the activities proposed to be carried out. Such charges should be shown separately in Annexure 4.3.2.
- 3 Any change in taxes after the notification of award of Contract will be adjusted accordingly in the payment structure.

2.11 Discount

The Bidders are advised not to indicate any separate discount. Unconditional discounts, if any, should be merged with the quoted prices. Discount of such type, indicated separately, will not be taken into account for evaluation purpose. However, in the event of such an offer, without considering discount, is found to be the lowest, UIDAI shall avail such discount at the time of award of Contract.

2.12 Bidder Qualification

The "Bidder" as used in the Bid documents shall mean the one who participate in the Bid process. The Bidder may be either the Constituted Attorney of the company or the Principal Officer or his duly Authorized Representative, in which case

- he/she shall submit a certificate of authority. All certificates and documents (including any clarifications sought and any subsequent correspondences) received hereby, shall, be furnished and signed by the Bidder.
- It is further clarified that the individual signing the Bid or other documents in connection with the Bid must certify whether he/she signs as:
 - a) Constituted Attorney of the Company.

OR

- b) The Principal Officer or his duly Authorized Representative of the company, in which case he/she shall submit a certificate of authority on behalf of the company duly authorized by the Board of Directors/organization.
- 3 The Bidder shall sign its Bids with the exact name of the company/organization to whom the Contract is to be issued.
- The Bidder shall clearly indicate their legal constitution and the person signing the Bids shall state his capacity and also source of his ability to bind the Bidder.
- 5 UIDAI may out rightly reject any Bid not supported by adequate proof of the signatory's authority.

2.13 Period of Validity of Bids

- Bids shall remain valid for at least 180 days from the date of opening of Technical Bids prescribed by UIDAI. A Bid valid for a shorter period may be rejected by UIDAI as non-responsive.
- In exceptional circumstances, UIDAI may ask the Bidder to extend the validity of the Bid. The validity of Bid Securing Declaration provided under Clause 2.3 shall also be suitably extended. However, a Bidder will not be permitted to modify its bid during the Bid validity period or the extended period, if any.

2.14 Revelation of Price

Price in any form or by any reason before opening the Commercial Bid should not be revealed, failing which the Bid shall liable to be rejected.

2.15 Terms and Conditions of Bidders

- 1 Terms and conditions of the Bidders will not be considered as forming part of their Bids.
- The Bidder should note that any deviation(s) or non-compliance will lead to rejection of Bid.

2.16 Local Conditions

- It will be imperative on each Bidder to fully acquaint himself with the local conditions and factors, which would have any effect on the performance of the Contract and / or the value.
- It will be imperative for each Bidder to fully inform themselves of all legal conditions and factors which may have any effect on the execution of the Contract as described in the bidding documents. UIDAI shall not entertain any request for clarification from the Bidder regarding such local conditions.
- It is the responsibility of the Bidder that such factors have properly been investigated and considered while submitting the bids and that no claim what so ever including those for financial adjustment to the Contract awarded under the bidding documents will be entertained by UIDAI and that neither any change in the time schedule of the Contract nor any financial adjustments arising thereof shall be permitted by UIDAI on account of failure of the Bidder to appraise themselves of local laws/conditions.

2.17 Headings

Headings of conditions hereto shall not affect the construction thereof.

2.18 Conditions for Pre-Qualification of Bidders

The copy of Bid shall be signed by the Bidder or a person or persons duly authorized to bind the Bidder to the Contract. The authorization shall be indicated by written Power-of-Attorney accompanying the Bid. All pages of the Bid, including un-amended printed literature, shall be signed and stamped by the person or persons signing the Bid.

The Bid shall contain no interlineations, erasures or overwriting, except as necessary to correct errors made by the Bidder, in which case such corrections shall be initiated by the person or persons signing the Bid.

The relevant sections shall be highlighted in the document attached.

#	Parameter	Pre- qualification Criteria Description	Evidence required
1.	Document	Bid Securing Declaration (BSD) is submitted in original	Bid Securing Declaration Form.

2.	Legal Entity/	Bidder shall be:	ICAI Registration Certificate, TAN,
	Registration of Firm	1. The Chartered Accountant Firm should be registered with The Institute of Chartered Accountants of India (ICAI) and have TAN, PAN, EPF, ESIC and GST Registration/Certificates. 2. The Firm should have been operational for at least 10 years after its registration as on 30th June 2025.	PAN, EPF, ESIC and GST Registration/ Certificates.
3.	Statutory Compliance Certificates	The Bidder must be registered with appropriate authorities for all applicable statutory duties/taxes and should comply with all taxation norms.	Self-certificate/Undertaking on Firm's letter head duly signed by authorized signatory that on the date of bid submission the Firm is complying with all statutory duties/taxation norms of India and possess all necessary certificates to this effect.
4.	Bidder's Turnover	The Bidder must have an average annual turnover of a minimum - INR 10 (Ten) Crore during the last three (3) financial years ending 31st March, 2024 in respect of providing professional services in the field of Audits, Taxation and Accounting related work.	Audited Financial Statement for the last three Financial Years (FY 2021-22, 2022-23, 2023-24) clearly mentioning turnover in respect of providing professional services in the field of Audits, Taxation and Accounting related work.
5	CAG Registration	The Firm should be empaneled with the Comptroller and Auditor General of India (C&AG) of India for FY 2025-26.	Empanelment/Registration Certificate with C&AG of India

6	Bidder's Experience (Projects)	The entity should have experience in handling Audit-related functions and/or GST compliances of at least three (3) Central/State Government organizations/PSUs/Statuary or Autonomous bodies/Nationalised Banks/Large Public or Private Listed Company during the last five years of overall value of Rs. 25 Lakhs each. Note: Cumulative value of the contracts with the same client/organization will be considered as a single work order	client certificates / completion certificates to be submitted.
7	Professional Strength	for the purpose of evaluation. 1. The Firm should have a minimum of 20 Nos. Chartered Accountants (Excluding partners) with itself as per ICAI Constitution Certificate as on the bid date. 2. The Firm should have at least Five (5) partners and at least Three (3) of them must be Fellow Chartered Accountants as per ICAI Constitution Certificate as on bid date.	ICAI Registration Certificate
8	Registered/ Branch Office	The Bidder should have a registered/branch office in Delhi - NCR.	ICAI Registration Certificate
9	Declaration on non- blacklisting	The bidder should not have been banned from carrying out business either with the Central Government Ministries /	Self-certificate/ undertaking on company's letter head duly signed by company's authorized signatory.

	Departments/PSUs or any	
	Autonomous/Statutory institution	
	during the last three financial	
	years and till the time of the	
	submission of the bid. A similar	
	ban after the submission of the Bid,	
	but before the award of the	
	Contract shall also disqualify the	
	Bidder.	
	1	

2.19 Last Date for Submission of Bids

- Bids should be uploaded on the Portal not later than the time and date specified in clause 1.2.1. After the prescribed time, the option for uploading of documents will not be there on the portal.
- 2 UIDAI at its discretion may extend the last date for the receipt of Bids, by amending the Bid Document, in which case all rights and obligations of UIDAI and Bidders previously subject to the last date will thereafter be subject to the last date as extended.

2.20 Modification and Withdrawal of Bids

- The Bidder may modify or withdraw its Bid after the Bids' submission on the portal, till last date of submission.
- No option will be there to modify the Bid, subsequent to the last date and time for receipt of Bids.
 - No Bid may be withdrawn in the interval between the last date and time for receipt of Bids and the expiry of the Bid validity period specified by the Bidder in the Bid. Withdrawal of a Bid during this interval shall result in the execution of Bidder's Bid Securing Declaration.

2.21 Address for Correspondence

All correspondence by UIDAI shall be made to the officer whose details are provided in the clause 4.1.1 of the Bid document.

2.22 Opening of Bids by UIDAI

- On the basis of information furnished in the Pre-Qualification Bid, Bidders will be evaluated for pre-qualification.
- 2 The Bids which do not qualify at this stage, will not be taken up for further evaluation.

2.23 Clarification

When deemed necessary, UIDAI may seek clarifications on any aspect from the Bidder. However, that would not entitle the Bidder to change or cause any change in the substance of the Bid submitted or value quoted.

2.24 Preliminary Examination

- UIDAI will examine the bids to determine whether they are complete, whether any computational errors have been made, whether the required Bid Securing Declaration (BSD) has been furnished, whether the documents have been properly signed, and whether the Bids are generally in order.
- Arithmetical errors will be rectified on the following basis. If there is a discrepancy between the unit cost and the total cost that is obtained by multiplying the unit cost and quantity, the unit cost shall prevail and the total cost shall be corrected. If the Bidder does not accept the correction of the errors, its Bid will be rejected. If there is discrepancy between words and figures, the amount in words will prevail.
- A Bid determined as not substantially responsive will be rejected by UIDAI and may not subsequently be made responsive by the Bidder by correction of the non-conformity.
- 4 UIDAI may waive any minor infirmity or nonconformity or irregularity in a Bid which does not constitute a material deviation, provided such waiver does not prejudice or affect the relative ranking of any Bidder.

2.25 Contacting UIDAI

- No Bidder shall contact UIDAI to influence the bidding process or on any matter relating to its Bid, from the time of the Bid opening till the time the Contract is awarded.
- Any effort by a Bidder to influence UIDAI's Bid evaluation, Bid comparison or Contract award decisions may result in rejection of the respective Bidder Bid.

2.26 Post Qualification

- UIDAI will determine to its satisfaction whether successful Bidder is qualified to perform the Contract satisfactorily.
- 2 This determination will take into account the Bidder's financial, technical,

implementation and post-implementation capabilities. It will be based upon an examination of the documentary evidence submitted by the Bidder as per document sought in the bid document as well as such other information as UIDAI deems necessary and appropriate.

An affirmative determination will be a pre-requisite for award of the Contract to the Bidder. A negative determination will result in rejection of the Bidder's Bid, and in such event UIDAI may proceed to the next lowest Bidder to make a similar determination of that Bidder's capabilities to perform satisfactorily or may call for new Bids.

2.27 Criteria for Evaluation of Bids

- 1 QCBS method of selection of Bidder will be implemented. The evaluation will be done in 3 phases by the Evaluation Committee (EC) setup by UIDAI.
- To meet UIDAI's requirements, as spelt out in the Bid Document, the successful bidder/Vendor must have the requisite, the technical know-how, and the financial ability that would be required to provide the Services sought for by UIDAI, for the entire period of the Contract. The evaluation process of the Bid proposed to be adopted by UIDAI is indicated below so as to provide the Bidders an idea of the evaluation process that UIDAI may adopt. Any time during the process of evaluation, UIDAI may seek specific clarifications from any or all Bidders.

3 Phase I: Evaluation of Pre - Qualification and Technical Bids

The Evaluation Committee (EC) shall first evaluate the Pre- Qualification Proposal as per the Pre – Qualification Criteria detailed in Clause 2.18. The Pre-Qualification proposal shall be evaluated based on the information provided in the Standard Forms (Section IV) and the supporting documents.

4 Phase II: Evaluation of Technical Bids

Evaluation of Technical bid is subject to meeting all pre-qualification criteria. A detailed evaluation will be subsequently carried out by UIDAI. In this phase, the Technical Bids will be reviewed for compliance of the Bid requirements with reference to the responses to the technical requirements given in Section IV and schedule of requirements of this Bid Document (Section V). Technical Bids which meet the below mentioned criteria will be eligible for consideration in the subsequent rounds. If required, UIDAI may seek specific clarifications from any or all Bidder(s) at this stage. UIDAI shall determine the Bidders that qualify for the next

phase after reviewing the clarifications provided by the Bidder(s). Technical Bid shall also contain unpriced BoQ as per clause 4.3.2 (without any cost mentioned).

Evaluation Committee while evaluating the Technical Proposals will have no access to the Financial Proposals until the technical evaluation is concluded, and the competent authority accepts the recommendation.

Bidders, if asked for, shall make presentation to Evaluation Committee. The qualification of the Bidder and the evaluation criteria for the Technical Proposal shall be as defined below. Each responsive proposal will be given a technical score. A Proposal shall be rejected at this stage if it does not respond to important aspects of the RFP, and particularly the Scope of Work or if it fails to achieve the minimum technical score mentioned below.

The criteria, parameters and point system for the detailed evaluation of Technical Proposals of the bidders shall be as follows :

Sl.	Parameters	Max. Marks
1	No. of Chartered Accountants associated with the firm (Excluding partners) as per ICAI constitution certificate as on 31.03.2025 20 to 35 Chartered Accountants - 70% 36 to 50 Chartered Accountants - 85% More than 50 Chartered Accountants - 100%	15 Marks
2	No. of Partners in the firm as per ICAI constitution certificate as on 31.03.2025 05 to 07 Partners - 70% 08 to 10 Partners - 85% 11 and above Partners - 100%	15 Marks
3	Average annual Turnover of the Firm for the last three financial years ending 31.03.2024 Rs. 10 Crore to 25 Crore- 70% More than Rs. 25 Crore to 40 Crore- 85% More than 40 Crores- 100%	15 Marks
4	Experience in handling Audit-related functions and/or GST compliances at Central/State Government organizations/ PSUs/Statuary or Autonomous bodies/ Nationalised Banks/Large Public or Private Listed Company during the last five years ending 31.03.2025:	15 Marks

	 a. 7.5 Marks for each project/work order of at least 1 Crore. b. 5 Marks for each project/work order at least 50 Lakhs. c. 2.5 Marks for each project/work order of at least 25 Lakhs. Note: Cumulative value of the contraction order for the purpose of evaluation. 	with overall value of er with overall value		
	Technically qualified professionals of	on the rolls of the		
	bidder with the following certificati			
5	Institute of Chartered Accountants of	India: -	10 Marks	
3			10 Mai KS	
	a. Internal Audit Certification	: 2 Marks		
	b. GST Certification	: 2 Marks		
	c. FAFD Certification	: 2 Marks		
	d. DISA Certification	: 1 Marks		
	e. CISA Certification	: 1 Marks		
	f. Certificate in IndAS/IFRS	: 1 Marks		
	g. Any Certification on Accounts/	4.34		
	Audit related function outside India	: 1 Marks	E Ml	
6	Firm having registered Head Office at De	Ini - NCR	5 Marks	
7	Solvency Certificate:- A Solvency Certificate	icate duly issued by		
	a scheduled commercial bank for the per	riod :-	5 Marks	
	- for the last three years & above as on 3			
	- for the last four years & above as on 30			
	- for the last 5 years and above as on 30.0	06.25 - 5 Marks		
	Total of Technical Marks (X):**		80 Marks	
	** The threshold Score for qualifying in the technical evaluation is minimum 56			
mark	as out of 80."			

^{**} Top 15 technically qualified scores in the technical evaluation (As per 'X' above) will be called for the presentation. The Bids of the Bidder who did not turn-up for the presentation will be summarily rejected. The presentation will be evaluated, and presentation marks (Y) will be obtained on a maximum Score of 10.

Sl.	Parameters	Max. Marks
1.	Presentation on: Approach, Methodology, List of key personnel, Strength of Firm and Experience of working with	10 Marks
	Govt./PSUs/ Autonomous Body / Statutory Body etc.	

2.	Proposed manpower's educational qualification, total years of experience, years of relevant experience, training & certifications etc., required under the scope of work (Refer Clause No. 4.2.4 and Clause No. 5.3)	10 Marks
	Total Presentation Marks (Y):	20 Marks
	Grand Total of Technical Marks (X+Y)	100 Marks

The total technical score /marks including presentation by the qualified Bidders (who are called for and attended presentation) shall be computed using following formula:

Total Technical Score (TBS) = X + Y

The technical scores of all other Bidders who are invited for the presentation and attended will be considered technically qualified and Financial Bids of only those Bidders will be opened.

Note: The Technical Bids shall be opened on the scheduled opening date and the Financial Bids of only those Bidders, who are found qualified in evaluation of the Technical Bids, shall be opened on later date (working day) with intimation to the Bidders.

Bidders who are declared technically qualified will be allowed to participate in the Financial Bid process.

5. Phase III: Evaluation of Commercial Bids

Evaluation of Price Bids:

The Commercial Bids shall be evaluated by the Purchaser for completeness and accuracy. Arithmetical errors will be rectified on the following basis. If there is a discrepancy between the unit price and the total price that is obtained by multiplying the unit price and quantity, the unit price shall prevail and the total price shall be corrected. If there is a discrepancy between words and figures the amount in words shall prevail.

Based on the Evaluation Criteria mentioned in the following clause, Financial Bid Score (FBS) shall be assigned to each Bid.

FBS= $100 \times (FL/F)$; Where:

F is the Total Bid Price quoted in the Bid under consideration

FL is the value of the lowest Commercial Bid

Overall Evaluation of Bids (TBS + FBS):

The overall valuation of Bids shall be done using QCBS method. The final score (FS) shall be calculated based on the formula:

Final Score (FS) = 80% of (TBS) + 20% of (FBS)

UIDAI will award the Contract to the successful Bidder whose Bid has been determined to be technically qualified and achieving the highest combined technical and financial score (FS), provided further that the Bidder is determined to be qualified to perform the Contract satisfactorily.

Tie-Breaking clause:

In case two or more Bidders score equal marks in the overall Bid Evaluation Score then the following Criteria will be adopted for tie-breaking in order of merit:

- (i) Annual value of turnover: Bidder having larger average annual turnover during the last three Financial Years ending on 31st March 2024 will be given preference.
- (ii) In case the average annual value of turnover is also same, then the bidder having executed similar works of larger values will be given preference.

2.28 Modification in Scope of Contract

The Parties may mutually agree in writing to modify the terms of the contract at any time during the period of Contract.

UIDAI shall reserve the right, not to hire all services quoted by the Bidder in this Bid.

2.29 UIDAI's Right to Accept Any Bid and to Reject Any or All Bids

UIDAI reserves the right to accept any Bid, and to annul the Bid process and reject all bids at any time prior to award of Contract, without thereby incurring any liability to the affected Bidder or Bidders or any obligation to inform the affected Bidder or Bidders of the grounds for UIDAI's action.

2.30 Notification of Award (NOA)

- UIDAI will notify through Notification of award (NOA) the successful Bidder in writing by letter or by email, to be confirmed in writing by letter, that its bid has been accepted. Within 10 days of receipt of this NOA, the bidder has to render his acceptance of notification of award.
- 2 Upon the successful Bidder's furnishing of performance security pursuant to Clause 2.32, UIDAI will promptly notify each unsuccessful Bidder and will discharge their Bid Securing Declaration.

2.31 Signing of Contract

- 1. At the same time as UIDAI notifies NOA the successful Bidder that its Bid has been accepted, UIDAI will send the Bidder the contract form (Appendix A of Section 6.1 provided in the Bid Document, incorporating all agreements between the parties).
- 2. Within 15 days of receipt of the contract form, the successful Bidder shall sign and date the Contract and return it to UIDAI.

2.32 Performance Bank Guarantee (PBG)

- 1 Within 15 days of the receipt of notification of award of the Contract from the Purchaser, the successful Bidder/vendor shall furnish the performance bank guarantee ("PBG") as per format prescribed in Appendix B of clause 6.2.
- 2 Failure of the successful Bidder/vendor to submit the PBG within 15 days shall constitute sufficient grounds for the annulment of the award and execution of Bid Securing Declaration (BSD), in which event UIDAI may call for new bids/ will proceed to the next lowest Bidder.
- 3 Performance Bank Guarantee will be for a total amount equivalent to 3% of total contract value and having validity till T0+3 years and 60 days. (T0= is the date of execution of Contract).
- PBG shall be invoked/encashed by UIDAI in the event the successful bidder/vendor:
 - i. Fails to perform the responsibilities and obligations as set out in the RFP and the Contract to the complete satisfaction of UIDAI,
 - ii. Misrepresentations of facts/information submitted to UIDAI.
- The PBG may be discharged/returned by UIDAI upon being satisfied that there has been due performance of the obligations of the Bidder under the Contract. However, no interest shall be payable on the PBG.
- In the event of the Bidder being unable to service the Contract for whatever reason, UIDAI would invoke/encash the PBG. Notwithstanding and without prejudice to any rights whatsoever of UIDAI under the Contract in the matter, the proceeds of the PBG shall be payable to UIDAI. UIDAI shall notify the bidder in writing of the exercise of its right to receive such compensation within 30 days, indicating the contractual obligation(s) for which the bidder is in default.
- 7 UIDAI shall also be entitled to make recoveries from the Bidder's bills, performance bank guarantee, or from any other amount due to him, an equivalent value of any payment made to him due to inadvertence, error, collusion, misconstruction, or misstatement.
- 8 In case the project/Assignment is delayed beyond the project schedule as mentioned in clause 4.2.3, the PBG shall be accordingly extended by the Bidder.
- On satisfactory performance and completion of the order in all respects and duly certified to this effect by the Project Coordinator, Contract Completion Certificate will be issued and the PBG will be returned to the Bidder, after 60 days of completion of contractual obligations.

2.33 Confidentiality of the Document

- The Successful bidder is required to furnish an undertaking that anything contained in their Bid Proposal shall not be disclosed in any manner by submitting NDA as per Appendix D.
- 2 Except with the written consent of the Purchaser, the Bidder shall not make use of any information supplied by the Purchaser for purposes of the Bidder or any specifications or other details mentioned in above clause otherwise than for the purpose of provision of services under the Contract and the Bidder shall not use any such information provision of similar services or part thereof for any other purpose.
- 3 NDA to be submitted by successful bidder.

2.34 Rejection Criteria

1 Technical Rejection Criteria

The following vital technical conditions should be strictly complied with, failing which the Bid will be rejected:

- i. Only the Bidders who quote for the complete Scope of Work and providing Services as indicated in this Bid Document, addendum thereof (if any) and any subsequent information given to the Bidder shall be considered. Incomplete bids will be rejected outright. Evaluation will be carried out for the total Scope of Work covered in the Bid document.
- ii. If the information provided by the Bidder is found to be incorrect/ misleading at any stage / time during the bidding process, the Bid will be rejected and Bid Securing Declaration will be executed.
- iii. Bid Securing Declaration will be executed in case of any breach of confidentiality clause 2.33 of this RFP.

2 <u>Commercial Rejection Criteria</u>

The following vital commercial conditions should be strictly complied with failing which the bid will be rejected:

- Any commercial quote, if found in any part of the Bid other than Commercial Bid, will lead to disqualification of Bidder and his Bid shall be rejected outright.
- ii. Bids of following kinds will be rejected:
 - a) Bids made without Bid Securing Declaration.
 - b) Bids which do not confirm unconditional validity of the Bid for 180 days from the date of opening of Technical Bid.
 - c) Bids where prices are not firm during the entire duration of the Contract

and / or with any qualifications.

- d) Bids which do not conform to UIDAI's price bid format.
- e) Bids which do not confirm to the completion period indicated in the Bid Document.
- iii. Bidder shall bear, within the quoted rates, all the taxes and duties as levied on them.

2.35 Fraud and Corruption

It is required that Bidders participating in the project adhere to the highest ethical standards, both during the selection process and throughout the execution of a Contract. The Purchaser:

- **a.** defines, for the purpose of this paragraph, the terms set forth below as follows:
- (i) "Corrupt practice" means the offering, giving, receiving, or soliciting, directly or indirectly, of anything of value to influence the action of a public official or members of the Evaluation Committee, in the selection process or in contract execution;
- (ii) "Fraudulent practice" means a misrepresentation or omission of facts in order to influence a selection process or the execution of a contract;
- (iii) "Collusive practices" means a scheme or arrangement between two or more Bidders with or without the knowledge of the Purchaser, designed to establish prices at artificial, non-competitive levels;
- (iv) "coercive practices" means harming or threatening to harm, directly or indirectly, persons or their property to influence their participation in a procurement process, or affect the execution of a contract;
- **b.** will reject a Proposal for award, if it determines that the Bidder recommended for award has, directly or through an agent, engaged in corrupt, fraudulent, collusive or coercive practices in competing for the contract in question;
 - The Bidders shall be deemed to be aware of the provisions on fraud and corruption stated in the specific clauses in the General Conditions of Contract.

3 Section III - General Conditions of Contract

3.1 Definitions

In this Contract, the following terms shall be interpreted as indicated:

- "UIDAI" means the Unique Identification Authority of India. 1.
- 2. "The Purchaser/Purchaser" means the Unique Identification Authority of India acting through CEO or any other representative authorized by the Unique Identification Authority of India.
- "Purchase Officer" means the officer signing the acceptance of Bid and 3. includes any officer who has the authority to execute the relevant Contract on behalf of the Purchaser.
- 4. The "Vendor" means the person or the firm or the company with whom the order of or the procurement of the Services is placed and shall be deemed to include the Vendor's successors, representatives (approved by the Purchaser), heirs, executors, administrators and permitted assigns, as the case may be, unless excluded by the terms of the Contract.
- 5. "Applicable Law" means the laws and any other instruments having the force of law in the Purchaser's country, as they may be issued and in force from time to time.
- 6. "Asset" means resources used by the Purchaser for carrying out its activities and shall include fixed assets such as land, buildings, plant machinery and equipment, vehicles, furniture and fixtures, office equipment etc. and current assets such as investments, inventories, sundry debtors, cash and bank balances, loans, advances, and other assets.
- 7. "Bidder" means any private or public entity who submits a bid in response to the request for a proposal (RFP).
- "Contract" means the Contract signed by the Parties and all the attached 8. documents i.e. the Sections like General Conditions Section (GC), Statement of Works Section, the Appendices and the Annexure of this RFP.
- "Contract Price" means the price to be paid for the performance of the 9. Services, in accordance with Clause 3.6 of Section III;
- 10. "Effective Date" means the date on which this Contract comes into force and effect pursuant to Clause 3.5 of Section III.

- 11. **"Foreign Currency"** means any currency other than the currency of the
- 12. Purchaser's country.
- 13. "GC" mean General Conditions of Contract.
- 14. **"Government"** means the appropriate Government of the Purchaser's country.
- 15. **"Local Currency"** means the currency of the Purchaser's country.
- 16. **"Party"** means the Purchaser or the Bidder, as the case may be, and "Parties" means both.
- 17. **"Personnel"** means persons hired by the Bidder and assigned to the performance of the Services or any part thereof.
- 18. **"Services"** means the work to be performed by the Bidder pursuant to this Contract.
- 19. **"Proposal/ Bid"** means the Pre-Qualification Proposal, Technical Proposal and Commercial Proposal submitted by the Bidder.
- 20. **"Pre-Qualification Proposal/Bid"** means the proposal submitted as per clause 4.1 of this RFP.
- 21. **"Technical Proposal/ Bid"** means the proposal submitted as per clause 4.2 of this RFP.
- 22. **"Financial/ Commercial Proposal Bid"** means the means the proposal submitted by Bidder as per clause 4.3 of this RFP.

3.2 Application

These General Conditions shall apply to the extent that they are not superseded by provisions in other parts of the Contract.

3.3 Use of Contract Documents and Information

- 1. The Vendor shall not, without the Purchaser's prior written consent, disclose the Contract, or any provision thereof, or any specification, plan, drawing, pattern, sample, or information furnished by or on behalf of the Purchaser in connection therewith, to any person other than a Personnel employed by the Vendor in the performance of the Contract. Disclosure to any such employed Personnel shall be made in confidence and shall extend only so far as it may be necessary for purposes of such performance.
- 2. The Vendor shall not, without the Purchaser's prior written consent, make use of any document or information enumerated in clause 2.4 except for purposes of performing the Contract.

3. Any document, other than the Contract itself, shall remain the property of the Purchaser and shall be returned (in all copies) to the Purchaser on completion of the Vendor's performance under the Contract, if so required by the Purchaser.

3.4 Delivery and Documents

Delivery of the services shall be made by the Vendor in accordance with the terms specified by the Purchaser in this RFP.

3.5 Period of Contract

- 1. This Contract shall come into effect on the date the Contract is signed by both Parties.

 The date the Contract comes into effect is defined as the Effective Date.
- 2. This Contract for "Agency for handling Internal Audit Functions in UIDAI" will be for three (3) years duration. It is extendable up to one year at the rate for 3rd Year mentioned in BoQ and on same terms and conditions of RFP with the mutual consent of both the parties.
- 3. Notwithstanding the provision of clauses above, the period of Contract shall be valid subject to the satisfaction of the Purchaser.
- 4. The Purchaser shall promptly notify the Vendor in writing of any issues in the licensing terms or upgrades and updates or miscellaneous issues. Upon receipt of such notice, the Vendor shall, with all reasonable speed, resolve the issue, without prejudice to any other rights which the Purchaser may have against the Vendor under the Contract.
- 5. If the Vendor, having been notified, fails to remedy the defect(s)/issue(s)within the period stated by Purchaser in such notification, the Purchaser may proceed to take such remedial action as may be necessary, at the Vendor's risk and expense and without prejudice to any other rights which the Purchaser may have against the Vendor under the Contract.
- 6. The Bidder shall begin carrying out the Services within 15 days from the signing of the Contract between UIDAI and the Bidder which would be considered as commencement of Services.
- 7. Penalty, as per the Service Level Agreement, will be applied, if the vendor is not able to deploy/provide resources as stipulated in the Delivery Schedule (Clause 4.2.3 of Section IV).

3.6 Payment schedule

The payments in respect of the Services shall be made as follows:

- a) The Bidder shall submit the invoice for monthly payment. The payment shall be released after making deductions on account of TDS and penalties for not meeting the SLAs.
- b) The payment shall be made only after receipt of status of actual manpower deployment, attendance record of manpower deployed duly certified by authorized signatory of Bidder and certificate of attendance certified by the accepting authority where the manpower is deployed has been submitted by the Bidder and approved as satisfactory by the "Purchaser" for that month.
- c) For the purpose of payment above, satisfactory means acceptance of the deliverables as well as acceptance of all the reports as mentioned above by the Purchaser after submission by the Bidder. It also includes time bound completion of activities that need to be done and agreed with UIDAI after award of Contract.
- d) If the invoices, reports, and deliverables submitted by the Bidder are not acceptable to the Purchaser, reasons for such non-acceptance should be recorded in writing; and in such event the Purchaser shall not release the payment due to the Bidder. Reports and deliverables not meeting the required quality or expectations or are incomplete shall not be accepted. In such case, the payment will be released to the Bidder only after it resubmits the reports and deliverable and which are accepted satisfactorily by the Purchaser.
- e) In case of early termination of the Contract, the pro rata payment shall be made to the Bidder for the period it has provided its Services to the Purchaser.

3.6.1 Payment terms for Manpower

- 1) Bidder shall quote yearly manpower charges.
- 2) Yearly charges will be converted to monthly charges for monthly payment.
- 3) The number of resources should not be reduced at any time during the currency of the Contract.
- 4) Any penalties and deductions will be done from monthly payments.
- 5) In case of any additional manpower is requested by UIDAI, the yearly charges for that year which is mentioned in BoQ will be apportioned appropriately i.e. yearly charges against a category and year in which request is made will be divided by number of resources in that year and then monthly rate will be calculated.

3.7 Currency of Payment

Payment shall be made in Indian Rupees only.

3.8 Change Orders

- 1. The Purchaser may at any time, by a written order given to the Vendor, make changes within the general scope of the Contract in any one or more of the following:
 - i) drawings, designs, or specifications.
 - ii) the place of delivery; or
 - iii) the Services to be provided by the Vendor.
- 2. If any such change causes an increase or decrease in the value, or the time required for, the Vendor's performance of any part of the work under the Contract, whether changed or not changed by the order, an equitable adjustment shall be made in the Contract Value or delivery schedule, or both, and the Contract shall accordingly be amended.
- 3. Increased cost quoted in the Bid for 2nd & 3rd year shall be applicable for only those resources who are initially deployed and continued with UIDAI after completing 1 year and 2 years respectively. If any replacement is deployed in place of a resource, payment for such resources shall be made as per the rates of 1st year.
- 4. The request for change of resource should be supported with Bio-data along with the certificates in respect of educational/ professional qualifications etc.

3.9 Contract Amendments

No variation in or modification of the terms of the Contract shall be made except by written amendment signed by the parties.

3.10 Assignment

The Vendor shall not assign, in whole or in part, its obligations to perform under the Contract, to other party, except with the Purchaser's prior written consent. The permission, if any, of the Purchaser must be taken within 15 days of award of the Contract.

3.11 Sub-contracts

- 1. No sub-contracting of the Services either in full or part is allowed.
- 2. The Purchaser shall select a single agency for the execution of this Contract and any sub-bidders, contractors, joint-ventures, partnerships or consortiums shall not be allowed.

3.12 Delays in the Vendor's Performance

- 1. Delivery of the manpower shall be made by the Vendor in accordance with the schedule specified by the Purchaser in clause 4.3.2.
- 2. An un-excused delay by the Vendor in the performance of its delivery obligations shall render the Vendor liable to any or all the following sanctions: forfeiture of its performance security, imposition of liquidated damages, and/or termination of the Contract for default.
- 3. If at any time during performance of the Contract, the Vendor should encounter conditions impeding performance of services, the Vendor shall promptly notify the Purchaser in writing of the fact of the delay, it's likely duration, and its cause(s). As soonas practicable after receipt of the Vendor's notice, the Purchaser shall evaluate the situation and may at its discretion extend the Vendor's time for performance, in which case the extension shall be ratified by the Parties. If the Vendor's request is not found acceptable to the purchaser, Clause 3.14 would be invoked.

3.13 Liquidated Damages

- 1. If the successful Bidder/Vendor fails to perform the Services within the time period(s) specified in the Contract, the Purchaser shall without prejudice to its other remedies under the Contract, deduct from the Contract Value, as liquidated damages as per Clause 5.5.3. Once the maximum is reached, the Purchaser may consider termination of the Contract pursuant to Clause 3.14.
- 2. The overall liquidated damages will be maximum of 10% of the total Contract value.
- 3. If the liquidated damages amount is more than the bill due to the Vendor, amount will be deducted from the subsequent bills. However, if the liquidated damages amount is not recoverable from the pending dues to the Vendor, in that case, same shall be recovered by encashing the PBG, if required.

3.14 Termination for Default

- 1. In case of any event of default on account of Vendor, the Purchaser shall provide notice of thirty (30) days to the Vendor to cure the event of default and meet the provision of Services.
- 2. Where an event of default subsists or remains uncured after 30 days of notice to vendor to resolve, the Purchaser may, without prejudice to any other remedy for breach of Contract, by written notice sent to the Vendor, terminate the Contract in whole or in part:
 - (i) If the Vendor fails to deliver any or all of the Services within the time period (s) specified in the Contract, or any extension thereof granted by the Purchaser;

OR

(ii) If the Vendor fails to perform any other obligation(s) under the contract;

OR

- (iii) If the Vendor becomes liable for maximum amount of liquidated damages under clause 3.13.
- 3. In the event the Purchaser terminates the Contract in whole or in part, the Purchaser may procure, upon such terms and in such manner as it deems appropriate, Deliverables/Services similar to those undelivered, and the Vendor shall be liable to the Purchaser for any excess costs for such similar Deliverables/Services including the efforts of the Purchaser for such arrangement. However, the Vendor shall continue performance of the Contract to the extent not terminated.

3.15 Force Majeure

1. Neither party shall bear responsibility for the complete or partial non-performance of any of its obligations (except for failure to pay any sum which has become due on account of performance of services under the provisions of the present contract), if the non-performance results from Force Majeure circumstances such as Flood, Fire, Earth-Quake and other acts of God as well as War, Military operation, blockade, pandemic situation or lockouts announced by the government due to pandemic, Acts or Actions of State Authorities or any other circumstances beyond the Parties control that have arisen.

- 2. In such circumstances the time stipulated for the performance of an obligation under the present contract is extended correspondingly for the period of action of these circumstances and their consequences, but not more than a period of 6 months.
- 3. The Party for which it becomes impossible to meet obligations under this contract due to Force Majeure conditions, is to notify in written form the other party of the beginning and cessation of the above circumstances immediately, but in any case, not later than 10 (Ten) days from the moment of their beginning.
- 4. Certificate of a Chamber of Commerce (Commerce and Industry) or other competent authority shall be a sufficient proof of commencement and cessation of the above circumstances.
- 5. If the impossibility of complete or partial performance of an obligation lasts for more than 6 (six) months, either Party hereto reserves the right to terminate the Contract totally or partially upon giving prior written notice of 30 (thirty) days to the other Party of the intention to terminate without any liability other than reimbursement on the terms provided in the Contract forthe Services received.

3.16 Termination for Insolvency

The Purchaser may at any time terminate the Contract without compensation to the Vendor, if the Vendor becomes bankrupt or otherwise insolvent, provided that such termination will not prejudice or affect any right of action or remedy which has accrued or will accrue thereafter to the Purchaser.

3.17 Termination for Convenience

The Purchaser may by written notice sent to the Vendor, terminate the Contract, in whole or in part at any time of its convenience by giving a prior written notice of sixty days. The notice of termination shall specify that termination is for the Purchaser's convenience, the extent to which performance of work under the Contract is terminated, and the date upon which such termination becomes effective.

3.18 Dispute Resolution

- a. The contract shall be governed by and constructed in accordance with the laws in India.
- b. In case any difference, dispute, controversy or claim which may arise between the Parties out of or in relation to or in connection with the contract, or the breach, termination, effect, validity, interpretation or application of the contract or as to their rights, duties or liabilities with respect to the contract including, either Party may send a written notice of dispute to the other Party (hereinafter referred to as the "Notice of Dispute") invoking Good Faith Negotiations (GFN). The Party receiving such Notice of Dispute shall respond to it, in writing, within 30 (thirty) days of the receipt of such Notice of Dispute. After the written response, both the Parties shall take measures to reach an amicable settlement and resolve the dispute through GFN within a period of 60 (sixty) days of the written response to the Notice of Dispute.
- c. In case the party receiving the Notice of Dispute fails to respond within 30 (thirty) days, or the dispute cannot be amicably settled by both the Parties within 60 (sixty) days of providing the written response, then such disputed matter shall resolve through arbitration in accordance with the provisions of the Arbitration and Conciliation Act, 1996 and Rules made there under, or any amendment made thereto. The reference shall be to a sole arbitrator, mutually appointed by the Parties. All fees, costs, or expenses pertaining to arbitration proceedings shall initially be borne equally by the Parties. The Parties shall have their dispute(s) resolved in terms of Section 29B Fast track procedure, of the Arbitration and Conciliation Act, 1996 (as amended). Any other fees and costs incurred by the Parties, other than in respect of arbitration proceedings, shall be borne by the respective Parties.
- d. The seat and venue of the arbitration shall be at New Delhi. The language of arbitration shall be English. The award given by the Sole arbitrator shall be final and binding on the Parties.
- e. The Parties shall continue to perform their respective obligations under the Contract despite the continuance of the arbitration proceedings, except for the disputed part under arbitration.

3.19 Language

The General terms and conditions/Contract shall be executed/written in the English language, which shall be the binding and controlling language for all matters relating to the meaning or interpretation.

3.20 Applicable Law

- a. That the Contract shall be governed by and construed in accordance with the Applicable Laws of India, including but not limited to the Aadhaar Act, 2016, the Indian Contract Act, 1872 etc.
- b. The courts in Delhi only shall have the exclusive jurisdiction to try, entertain and adjudicate any dispute pertaining to Contract.

3.21 Notices

- 1. Any notice by one Party to the other pursuant to the Contract shall be sent in writing or by email and confirmed in writing to the address specified for that purpose in the Contract.
- 2. A notice shall be effective when delivered or on the notice's effective date and time, whichever is later.

3.22 Deductions

Payments shall be subject to deductions of any amount, for which the Vendor is liable under the Contract against this Bid.

3.23 Taxes and Duties

The Vendor shall be entirely responsible for all taxes, license fees etc.,incurred until delivery of the Services to the Purchaser. If there is any reduction/increase in duties and taxes due to any reason whatsoever, after Notification of Award, the same shall be passed on to the Purchaser/vendor.

3.24 Continuing Support

The Vendor shall provide adequate and appropriate support and participation, on a continuing basis, in tuning and support all Vendor supplied Services to meet the requirements of the applications.

3.25 Conflict of Interest

The Bidder should not have any Conflict of Interest that may affect selection of Bidder during selection stage. The bidder shall disclose to the Purchaser in writing, all actual and potential conflicts of interest that exist, arise or may arise (either forthe bidder or the bidder's Team) in the course of performing the Services as soon as practical after

it becomes aware of that conflict. Failure to disclose such situations may lead to the disqualification of the bidder/vendor or termination of its contract during execution of the assignment.

3.26 Delivery and Risk Purchase

- 1. The time and the date of delivery of the Services as specified in Section V of this RFP shall be deemed to be the essence of the Contract and delivery must be completed as per delivery schedule as per schedule specified at clause 4.2.3.
- 2. The Purchaser reserves the right to terminate the Contract or a portion thereof and purchase the Services as specified in Section V of this RFP at the risk and cost of Vendor after giving due notice to the Vendor even before completion of the contractual delivery schedule if it becomes apparent that Vendor will not be able to fulfill the contractual obligations. In case the Vendor fails to provide the desired Services or a portion thereof within the contractual delivery schedule, the Purchaser has the right to procure the Services or a portion thereof at the risk and cost of Vendor.
- 3. In the event of termination of the Contract by Purchaser at the risk and cost of the Vendor, the Vendor shall be liable for any loss which the Purchaser may sustain on account of risk purchase but the Vendor shall not be entitled to any gain on such purchase made against such default. The manner and method of such purchase shall be at the entire discretion of the Purchaser, whose decision will be final.

3.27 Fall Clause

The following fall clause will form part of the Contract placed on successful Bidder: -

- a) The prices charged for Services provided under this Contract by the Bidder shall in no event exceed the lowest price at which the Bidder sells Services or offers to sell Services of identical description/similar buying circumstances to any persons/organizations including the Purchaser or any department of the Central or State Government or any statutory undertaking of the Central or State Govt., as the case may be, during the currency of the Contract.
- b) If any time during the contract period the Bidder reduces the sale price, sells or offers to sell such Services to any person/organization including the Purchaser or any department of State Govt or Central Govt. for statutory undertaking of the Central or State Govt. as the case may be at a price lower than the price chargeable under the Contract, he shall forthwith notify such reduction of sale or offer to sell to

the purchaser and the price payable under the contract for the Services supplied after the date of coming into force of such reduction or sale or offer to sell shall stand correspondingly reduced.

3.28 Indemnification and Limitation of Liability

3.28.1 General Indemnity

Subject to Clause 3.28.2 below, the successful Bidder/Vendor (the "Indemnifying Party") undertakes to indemnify the Purchaser and its nominated agencies (the "Indemnified Party") from and against all losses, claims, damages, compensation etc. on account of bodily injury, death or damage to tangible personal property arising in favour of any person, corporation or other entity (including the Indemnified Party) attributable to the Indemnifying Party's negligence, willful default, lack of due care or breach of terms of this Contract.

3.28.2 IPR Indemnity

If the Indemnified Party promptly notifies the Indemnifying Party in writing of third claim against the Indemnified Party party Deliverables/Services provided by the Indemnifying Party infringes a copyright, trade secret, patent or other intellectual property rights of any third party, the Indemnifying Party will defend such claim at its expenseand will pay any costs or damages that may be finally awarded against the The Indemnifying Party will not indemnify the Indemnified Party. Indemnified Party, however, if the claim of infringement is caused by (a) The Indemnified Party's misuse or modification of the Deliverables; (b) The Indemnified Party's failure to use corrections or enhancements made available by the Indemnifying Party; (c) The Indemnified Party's use of the Deliverables in combination with any product or information not owned or developed or supplied by the Indemnifying Party. If any of the Deliverables is or likely to be held as infringing, the Indemnifying Party shall at its expense and option either (i) procure the right for the Indemnified Party to continue using it, (ii) replace it with a non-infringing equivalent, (iii) modify it to make it non-infringing.

3.28.3 Conditions for Indemnity

Without prejudice to the rights of the Purchaser in respect of indemnification for any claim:

- 3.28.3.1 The Purchaser shall notify the Vendor upon receipt of any notice of claim setting out in reasonable particulars, the details of such notice of claim;
- 3.28.3.2 Immediately upon receipt of notification of any claim from the Purchaser, the Vendor shall within a period of 5 days from date of receipt of such notice from the Purchaser, notify the Purchaser whether the Vendor wish to assume the defence in relation to such claim (including settlement or resolution thereof). Thereafter, the Vendor shall be entitled in consultation with the Purchaser, and only to the extent such action does not in any manner compromise, prejudice or adversely affect the interests of the Purchaser, to take such action as mutually agreed upon by Vendor and the Purchaser to avoid, dispute, deny, resist, appeal, compromise or consent such claim, within a period of 30 days from the date of receipt of such claim notification;
- 3.28.3.3 Notwithstanding anything contained herein, the Vendor and the Purchaser agree and covenant that a notice by the Purchaser to the Vendor in relation to the claim as aforesaid shall amount to express acceptance and consent by the Vendor to indemnify the Purchaser for all losses in relation to such claim. Upon notice by the Vendor, the Purchaser shall reasonably co-operate with the Vendor at the sole costs of the Vendor, only to the extent the same does not in any manner compromise, prejudice or adversely affect the rights of the Purchaser. The Purchaser shall have the right, at its option, to participate in the defence of such claim;
- 3.28.3.4 If the Vendor fails to take any action as per the above clause within the time period as specified therein, the Purchaser shall have the right, in its absolute discretion, to take such action as it may deem necessary to avoid, dispute, deny, resist, appeal, compromise or contest or settle any claim (including without limitation, making claims or counterclaims against third parties). If the Vendor does not assume control of the defence of such claims (as mentioned above), the entire defence, negotiation or settlement of such claim by the Purchaser shall be deemed to have been consented to by, and shall be binding upon,

Vendor as fully as though the Vendor alone had assumed the defence thereof and a judgment had been entered into by the Vendor, for such claim in respect of the settlement or judgment.

3.28.4 Risk Purchase

If the Vendor fails to perform its obligations (or any part thereof) under this Contract or if the Contract is terminated by the Purchaser due to breach of any obligations of the Vendor under this Contract, the Purchaser reserves the right to procure the same or equivalent Services /Deliverables from alternative sources at the vendor's risk and responsibility. Any incremental cost borne by the Purchaser in procuring such Services/Deliverables shall be borne by the Vendor. Any suchincremental cost incurred in the procurement of the such Services/ Deliverables from alternative source will be recovered from the undisputed pending due and payable Payments /Security Deposit / Bank Guarantee provided by the Vendor under this Contract and if the value of the Services/Deliverables under risk purchase exceeds the amount of Security Deposit and / or Bank Guarantee, the same may be recovered, if necessary, by due legal process.

3.28.5 Limitation of Liability

- 1. Except in case of gross negligence or wilful misconduct on the part of the Bidder or on the part of any person or company acting on behalf of the Bidder in executing the work or in carrying out the Services, the Bidder, with respect to damage caused by the Bidder to property and/ or assets of the Purchaser or of any of the Purchaser's Bidders, shall not be liable to Purchaser:
 - a. For any indirect or consequential loss or damage; and;
 - b. For any direct loss or damage that exceeds
 - i. The Contract Value, or
 - ii. The proceeds the Bidder may be entitled to receive from any insurance maintained by the Bidder to cover such a liability, whichever of (i) or (ii) is higher.
- 2. This limitation of liability shall not affect the Bidder's liability, if any, for damage to third parties caused by the Bidder/ Bidder's personnel/ team or any person or firm/ company acting on behalf of the Bidder in executing the work or in carrying out the Services.

3.28.6 Allocation of liability

The allocations of liability represent the agreed and bargained forunderstanding of the Parties and compensation for the Services/ Deliverables reflects such allocations.

3.29 Other obligations of Vendor

- 1. Vendor shall ensure that the software's, if any, being offered as part of the Contract does not contain any kind of malicious code that would activate procedures to:
 - Inhibit the desired and the designed function
 - Cause physical damage to the user or his equipment during the operational exploitation of the equipment
 - Tap information regarding the system, network, network users and information stored on the network that is classified and/or relating to National Security, thereby contravening Official Secrets Act 1923.
 - Cause any loss or corruption of data

Vendor shall be responsible for ensuring that the infrastructure provides multiple levels, layers of security against any kind of snooping, break-in and that there are no Trojans, Viruses, Worms, Spywares or any malicious software, components or shipped software developed or deployed.

2. The vendor shall ensure compliance with the law of the land including the Aadhaar (Targeted Delivery of Financial and Other Subsidies, Benefits and Services) Act, 2016 and its regulations.

3.30 Termination for Misrepresentations of facts

- 1. In case if any misrepresentation of fact/information, corrupt or fraudulent practices by successful Bidder comes to the notice/information of UIDAI/Purchaser after signing of the Contract, in that case UIDAI/Purchaser shall provide notice of thirty (30) days to successful Bidder for providing the clarification.
- 2. Where issue is not resolved within 30 days of notice to successful Bidder/ or if the satisfactory clarification is not rendered by successful Bidder in 30 days in that case Purchaser may, without prejudice to any other remedy for breach of Contract, by written notice to the successful Bidder, terminate the Contract in whole or in part:
 - a) If the successful Bidder, in the judgment of UIDAI/Purchaser, has engaged in misrepresentation of facts/information, corrupt or fraudulent practices in competing for the Bid or in executing this Contract;

OR

- b) If the successful Bidder submits to UIDAI/Purchaser a false statement/ facts.
- 3. UIDAI/Purchaser may also forfeit the PBG and/or other necessary legal action.

3.31 Good Faith

The Parties undertake to act in good faith with respect to each other's rights under this Contract and to adopt all reasonable measures to ensure the realization of the objectives of this Contract.

3.32 Suspension

The "Purchaser" may, by written notice of suspension to the Bidder, suspend all payments to the Bidder hereunder if the Bidder fails to perform any of its obligations under this Contract, including the carrying out of the Services, provided that such notice of suspension

- i. shall specify the nature of the failure, and
- ii. shall allow the Bidder to remedy such failure, if capable of being remedied, within a period not exceeding thirty (30) days after receipt by the Bidder of such notice of suspension.

3.33 Intellectual Property Rights

- 1. The intellectual property rights to all the deliverables listed under shall remain sole and absolute property of the "Purchaser".
 - 2. The intellectual property rights of all the tools, processes, utilities, and methodology used in the provision of Services hereunder and/or development of any deliverables hereunder which were existing with Bidder prior to this Contract shall remain property of the bidder. All new ideas, inventions, innovations, or developments conceived, developed or made by Bidder or its consultants/employees ("Software Vendor Materials") under this Contract, shall remain the property of the Purchaser. Bidder grants the Purchaser a non-exclusive, non-transferable, worldwide, perpetual, irrevocable, paid up license to use the Software Vendor Materials to the extent the same have been incorporated in the deliverables or are required for the use of deliverables in terms of this Contract.

3.34 Documents Prepared by the Bidder to be the Property of the Purchaser

- 1. All plans, drawings, specifications, designs, reports, forms, other documents and software (including source code) submitted, developed or customized by the Bidder under this Contract shall become and remain the property of the Purchaser, and the Bidder shall, not later than upon termination or expiration of this Contract, deliver all such documents to the Purchaser, together with source code of the software so developed under this contract along with a detailed inventory thereof.
- 2. The Bidder may retain a copy of such documents, on prior permission of UIDAI, but shall not use anywhere, without taking permission, in writing, from the Purchaser and the Purchaser reserves right to grant or deny any such request. If license agreements are necessary or appropriate between the Bidder and third parties for purposes of development of any such computer programs, the Bidder shall obtain the "Purchasers" prior written approval to such agreements, and the "Purchaser" shall be entitled at its discretion to require recovering the expenses related to the development of the program concerned.

RFP FOR ENGAGEMENT OF A CA FIRM/COMPANY FOR HANDLING **INTERNAL AUDIT FUNCTIONS OF UIDAL**

Section IV - Contents of the Bids

4.1 Pre-qualification

4.1.1	Propos	al Subm	ission	Form
	o p o o .			

To:

The Director (Finance and Accounts) Unique Identification Authority of India (UIDAI) M/o Electronics & Information Technology, Govt. of India (GoI), 5th Floor, Bangla Sahib Road, Behind Kali Mandir, Gole Market, New Delhi – 110001

Date: _____

Dear Sir/Madam,
We, the undersigned, offer to provide services in accordance with your Request for
Proposal dated, and our Proposal. We are hereby submitting our
Proposal, which includes the Pre-Qualification Proposal, a Technical Proposal and
Financial Proposal as per e-Procurement three cover System.
We hereby declare that we are submitting our Proposal as a single bidder/ company
and have not formed or intend to form or execute the Contract with any other entity,
sub-contractors, or consortium.
We hereby declare that all the information and statements made in this Pre-
Qualification Proposal are true and accept that any misinterpretation/representation
contained in it may lead to our disqualification. We undertake, in case our Proposal is
accepted, to initiate the Services related to the assignment not later than
(Commencement date).
We understand that you are not bound to accept any Proposal you receive.We
remain,
Yours sincerely,
Authorized Signature [In full and initials]: Name and Title of Signatory:
Name of Company:
Address:
Location:

4.1.2 Bidder's Organization and Experience

A - Bidder's Organization

[Provide here a brief description of the background and organization of your company/entity and each associate company for this assignment. Also provide a brief description on the ownership details, date, and place of incorporation of the company, objectives of the company etc. Apart from this, also provide information on the Annual Turnover of the company for the last 3 financial years ending on 31st March 2024 as required in the form below]

Organization and Financial Information

Form A 1: Details of the Organization - Bidder		
Name		
Date of Incorporation		
Date of Commencement of Business		
Address of the Headquarters		
Address of the Registered Office in		
India		
Area of expertise with respect to this		
Project		
Contact details (name, address, phone		
no. and email)		

Form A 2: Financial Information - Bidder					
	FY 2021-22	FY 2022-23	FY 2023-24		
Annual Turnover (in INR Crores)					
Other Relevant Information					

B - Bidder Company's Experience

[Using the format below, provide information on each assignment for which your company was legally contracted either individually as a corporate entity or as one of the major companies within an association, for carrying out Services as requested under this assignment. Bidders should provide all the details (including documentary evidence like copy of work order/contract document/completion certificate from customer/certificate from auditor/other customer testimonials along with supporting certificate from auditors) as required in the form. In case, information required by UIDAI is not provided by Bidder, UIDAI shall proceed with evaluation based on information provided and shall not request the Bidder for further information. Hence, responsibility for providing information as required in this form lies solely with Bidder.]

Please provide citations of 3 (three) relevant completed skilled manpower for outsourcing Internal Audit functions and/or GST compliances in any Organization (Please refer clause 2.18).

Assignment name:	
Country:	
Location within country:	
Name of Purchaser:	
Address:	
Annual Turnover/Budget of the Purchaser:	
Start date (month/year):	
Completion date (month/year):	
Name of senior professional staff of your company involved & functions performed (indicate most significant profiles such as Project Director/Coordinator):	
Narrative description of Project:	
Description of actual services provided by your staff within the assignment:	
Approx. Value of Contract: (Mandatory field) (Mention contract value in INR)	
Duration of assignment (months):	
Total No. of staff- months of the assignment	
Approx. value of the services provided by	
your company under the contract (in INR):	
No. of professional staff-months provided by associated Bidders:	

Company's	Name:	

4.2 Technical Bid

4.2.1 Bid Particulars

A - Bidder's Organization

[Provide here a brief description of the background and organization of your company/ entity for this assignment. Also provide a brief description on the ownership details, date, and place of incorporation of the company, objectives of the company etc.]

B - Bidder Company's Experience - Skilled Manpower for Internal Audit and GST Compliances functions.

[Using the format below, provide information on each assignment for which your company was legally contracted either individually as a corporate entity or as one of the major companies within an association, for carrying out Services as requested under this assignment. Bidders should provide all the details (including documentary evidence like copy of work order/contract document/completion certificate from customer/certificate from auditor/ other customer testimonials including video along with supporting certificate from auditors) as required in the form. In case, information required by UIDAI is not provided by Bidder, UIDAI shall proceed with evaluation based on information provided and shall not request the Bidder for further information. Hence, responsibility for providing information as required in this form lies solely with Bidder.]

Assignment name:	
Country:	
Location within country:	
Name of Purchaser:	
Address:	
Annual Turnover/Budget of the	
Purchaser:	
Start date (month/year):	
Completion date (month/year):	
Name of senior professional staff of your company involved and functions performed (indicate most significant profiles):	

Approx. Value of Contract:		
(Mandatory field)		
(Mention contract value in INR values :)		
Duration of assignment (months):		
Total No. of staff-months of the assignment:		
Approx. value of the services provided by your company under the contract (in INR):		
No. of professional staff-months provided by associated Bidders:		
Narrative description of Project:		
Description of actual services provided by your staff within the assignment:		

Company's Name:_____

4.2.2 Technical Bid Letter

To:

The Director (Finance and Accounts)
Unique Identification Authority of India (UIDAI)
M/o Electronics & Information Technology, Govt. of India (GoI),
5th Floor, Bangla Sahib Road, Behind Kali Mandir,
Gole Market, New Delhi – 110001

Rof ·	Bid 1	Nο			
Rei :	DIU	NU.			

Sir,

We declare that we are the service providers of _____

2. We hereby offer to supply the Services at the cost mentioned in Commercial Bid.

3. PERIOD OF DELIVERY

We do hereby undertake that, in the event of acceptance of our Bid, the provision for Services shall be made as stipulated in the Schedule to the Bid Document as given in **Annexure 4.2.3.**

4. TERMS OF DELIVERY

The prices quoted are inclusive of all charges wherever required.

- **5.** We enclose herewith the complete **Technical Bid** as required by you. This includes duly filled and signed **Section IV**.
- **6.** We agree to abide by our offer for a period of **180 days** from the date of opening of the technical bids.
- 7. We have carefully read, understood and accept the terms & conditions specified in the Bid. We do hereby undertake to services as per these terms and conditions of the bid document.
- **8.** Certified that the Bidder is:
 - a) The Constituted Attorney of the Company and the person signing the Bids is the constituted attorney of the Company.

OR

b) The Principal Officer or his duly Authorized Representative of the Company, and he has the authority to refer to arbitration disputes concerning the business of the Company by virtue of the general power of attorney.

(**NOTE:** Delete whatever is not applicable.)

- 9. Bid Securing Declaration Form, in original, is enclosed in the cover containingPre-Qualifying and Technical Bid.
- **10.** We do hereby undertake, that, until a formal contract is prepared and executed, this Bid, together with your written acceptance thereof and placement of Letter of Intent awarding the Contract, shall constitute a binding contract between us.

Dated :,	Sign & seal of the bidderName :
	Complete Address :
	Telephone/Mobile No. :
	Fax No:

Details of enclosures:

- 1.
- 2.
- 3
- 4.

4.2.3 Delivery Schedule

S No	Activity/ Deliverable	Timeline for completion*
1	Project Initiation	This is the date of commencement of contract (T0)
2	Reporting of all manpower	T0 + within 15 days
3	Conducting of Internal Audits	As per Agreed schedule
4	Submission of draft Internal Audit Report	Within two weeks of completion of Internal Audit
5.	GST Reconciliations/Audit	Within one month of respective due dates of filing GST Returns

^{*} The timeline mentioned for Acceptance by purchaser is tentative. This may increase or decrease for which UIDAI shall not be responsible for any cost if incurred by vendor on this account.

Signature of Bidder

Date

Place

4.2.4 Brief profile of manpower initially proposed to be deployed for execution of contract

The brief profile of manpower proposed to be part of initial team providing services covered under the Scope of Work of this RFP shall be structured as under:

Location	Role	Educational Qualification	Year of experience	Nature of experience	Current & previous organizations, where worked
UIDAI HQ	Project Manager				
	Deputy Manager – Internal Audit				
New Delhi	Deputy Manager – GST Compliance				
	Senior Audit Executive				

4.3 Commercial Bid

4.3.1 Commercial Bid Letter

Го:
The Director (Finance and Accounts) Unique Identification Authority of India (UIDAI), M/o Electronics & Information Technology, Govt. of India 5th Floor, Bangla Sahib Road, Behind Kali Mandir Gole Market New Delhi - 110001
Dear Sir/Madam,
We, the undersigned, offer to provide <i>services</i> in accordance with your Request for Proposal dated
We hereby declare that all the information and statements made in this Proposal are true and accept that any misinterpretation contained in it may lead to our disqualification.
We undertake that, in case our Proposal is accepted, to initiaterelated to the assignment not later than the date
We understand that you are not bound to accept any Proposal you receive.
Yours sincerely,
Authorized Signature [In full and initials]:
Name and Title of Signatory:
Name of Company:
Address:
Location:
Date:

4.3.2 Cost of Services Offered

Annexure 4.3.2

This form shall be used for arriving at the lump-sum value of the contract. The form shall include the staff-rates for ALL the resources proposed to be deployed during the entire duration of the project.

SI. No.	Manpower Description	No. of Manpower	Year 1 Cost in Rs (Excl. taxes)	Year 2 Cost in Rs (Excl. taxes)	Year 3 Cost in Rs (Excl. taxes)	Taxes in %	Total Amount in Rs. for period of Contract i.e. 3 years (Excl. taxes)	Total Amount in Rs. for period of Contract i.e. 3 years (Incl. Taxes)
			a	b	С	d	e = a+b+c	f = e * (1+d/100)
1	Manpower Cost							
1.1	Project Manager	1						
1.2	Deputy Manager – Internal Audit	1						
1.3	Deputy Manager - GST Compliance	1						
1.4	Senior Audit Executive	1						
GRAND TOTAL								

Note:- Increased cost quoted in the Bid for 2nd & 3rd year shall be applicable for only those resources who are initially deployed and continued with UIDAI after completing 1 year and 2 years respectively. If any replacement is deployed in place of a resource, payment for such resources shall be made as per the rates of 1st year.

5 Section V – Scope of Work

5.1 Introduction

- 1. This RFP aims to identify and engage an agency for handling Internal Audit functions and ensuring GST Compliance at the Headquarter, Regional Offices (including State Offices) and Tech Centre of UIDAI.
- 2. Before its establishment as a statutory authority, Unique Identification Authority of India (UIDAI) was functioning as an attached office under erstwhile Planning Commission. Subsequently, vide gazette notification dated 12th July 2016, Govt. of India established, the Unique Identification Authority of India as a statutory body under the Ministry of Electronics and Information Technology, to exercise the powers conferred on, and to perform the functions assigned to it under the Aadhaar Act, 2016.
- 3. The head office of UIDAI is located at New Delhi, with eight regional offices at Bengaluru, Hyderabad, Lucknow, Guwahati, New Delhi, Ranchi, Mumbai and Chandigarh and offices for Central Identities Data Repository operations at Bengaluru and Manesar.
- 4. The Finance & Accounts Division (FD) of UIDAI is headed by the Deputy Director General (Finance), who is the Financial Advisor in UIDAI. The function of the Division includes screening of proposals involving financial implications, rendering concurrence and advice on financial matters, preparation of budget, expenditure and cash management, drawing and disbursement of funds, releasing payments to staff and vendors, book keeping and preparation of accounts, internal audit, disbursal of pay & allowances, etc. The details are available on UIDAI's website.
- 5. The average annual budget of UIDAI is in the range of Rs 1,500 crore. It is further estimated that number of financial transactions at Headquarters is 10,000 per annum whereas at each RO estimated number of transactions will be 1000 per annum. More than 90% of the total expenditure of UIDAI takes place at Headquarters.

6. At present UIDAI receives funds mainly from the Central Government in the form of Grants. Additionally, periodic fees from the agencies using Aadhaar services like Authentication Charges, License Fees, and miscellaneous receipts such as tender fee, penalties, security deposits etc. also form the part of UIDAI's receipt.

5.2 Resource Engagement

- 1. The Agency will provide skilled resources with qualification and experience detailed in Para 5.3 for carrying out activities mentioned in Para 5.4 of scope of work.
- 2. The requirement of manpower to carry out tasks mentioned in this scope of work shall be assessed by the vendor. However, UIDAI shall insist on providing a minimum number of resources during the period of Contract, which would be as under:

 One Project Manager, One Deputy Manager Internal Audit, One Deputy Manager –
 GST Compliance and One Senior Audit Executive at UIDAI Head Quarter.
- 3. The cost of requirement of manpower, in addition to those stipulated as minimum required, for performing obligations specified in this scope of work shall be borne entirely by the vendor and without charging any extra cost to the Purchaser.
- 4. The manpower provided by the vendor under this contract shall be used specifically for the functions as specified in this scope of work.
- 5. No additional cost shall be charged by the vendor falling under the scope of this RFP, but cost of which are not directly covered under the Bill of Quantities/ Commercial Bid.
- 6. Frequency of conducting Internal Audits alongwith Travel Entitlements are as per Appendix E of this RFP.

5.2.1 Description of Personnel

- a) The title, job description, minimum qualification, and estimated period of engagement in carrying out of the Services of each of the Bidder's Personnel are as per the Bidder's Proposal to provide the Services.
- b) The Bidder must provide sufficient number of key personnel to maintain the services delivered and do the necessary ramp up that may be required during the Contract duration.

- c) The Bidder must always provide such minimum number of Personnel as stipulated under Schedule of Requirement, Scope of Work.
- d) Any request for addition of resource shall be met in 30 days. Similarly, 30 days' notice will be given for decrease in number of resources or replacement of a resource.
- e) Hardware requirements for staff: UIDAI will provide Desktops/Laptops to onboarded resources as per extant UIDAI Policy.

5.2.2 Approval of Personnel

In respect of Personnel which the Bidder proposes to use for carrying out of the Services, the Bidder shall furnish necessary details and the tentative engagement plan at the time of signing the Contract.

The Bidder shall obtain the Purchaser's prior approval in writing before any change or addition to the Personnel listed as mentioned by bidder after the signing of Contract.

5.3 Qualification of resources

The minimum educational qualification and professional experience of the resources to be provided under the RFP shall be as under:

• Project Manager:

- Essential Qualification: Should be a qualified Chartered Accountant from Institute of Chartered Accountants' of India with at least ten years of post-qualification experience in handling Audits related functions.
- Desirable: Master of Business Administration in Finance. Persons having experience of working with Government Organizations/Public Undertakings /Statutory Bodies/Autonomous Bodies will be given preference. He/she should have a quality to oversee the entire Internal Audits functions and GST compliances of UIDAI.

• Deputy Manager - Internal Audit:

Essential Qualifications: Should be a qualified Chartered Accountant from Institute of chartered Accountants of India or Cost Works Accountant from the Institute of Cost Accountants of India with at least five years of postqualification experience in handling Audit related functions and should have a good knowledge of developing Internal Audit framework, planning and

monitoring of Audits.

 <u>Desirable</u>: Persons having experience of working with Government Organizations/Public Undertakings /Statutory Bodies/Autonomous Bodies and proficient in "ERP-Tally Accounting Software" will be given preference.

• Deputy Manager - GST Compliance:

- Essential Qualifications: Should be a qualified Chartered Accountant from Institute of chartered Accountants of India or Cost Works Accountant from the Institute of Cost Accountants of India with at least five years of postqualification experience in handling GST related matters including but not limited to Directorate General of GST Intelligence (DGGI) or Anti Evasion division matters.
- Desirable: Persons having experience of working with Government Organizations/Public Undertakings /Statutory Bodies/Autonomous Bodies and handled GST Litigations at Adjudication/Tribunal/High Court level will be given preference.

Senior Audit Executive :

- <u>Essential Qualification</u>: Bachelor's degree in Commerce with at least five years' post qualification experience in Accounts, Audit, Taxation and other related functions. The candidate must be Conversant/proficient in "ERP-Tally Accounting Software"
- <u>Desirable</u>: CA (Inter) /ICWAI (Inter) /MBA (F)/ Post graduate degree in Commerce. Persons having experience of working in a Government Organizations/Public Undertakings/ Statutory Bodies/Autonomous Bodies will be given preference.

5.4 Manpower Service.

5.4.1 Internal Audit functions at UIDAI

Checklist of issues to be examined during Internal Audit of offices of UIDAI:

1. Compliance/Regularity issues:

This domain comprises review and analysis of records with a view to ensuring adherence to extant rules, regulations, UIDAI policies, processes and procedures and would involve financial and non-financial issues. The illustrative list of financial and non-financial issues to be reviewed under this domain, including an evaluation of the relevant underlying internal controls are as follow:

1.1 Financial issues:

- I. Regulation of pay and allowances as per extant rules and regulations.
- II. Regulation of entitlements (LTC, TA, Medical reimbursement, leave salary, Pension contribution etc) as per extant rules and regulation.
- III. Regulation of other claims/benefits as per entitlements.
- IV. Regulation of payments to various agencies towards procurement of goods and services such as Housekeeping, Security arrangements, Vehicle hiring, procurement of hardware/software, outsourced manpower, engagement of consultants/professionals, miscellaneous procurements, etc, in accordance with the Terms and conditions of the contract/terms of Purchase orders. This would also involve review of applications of SLAs/levy of Liquidated damages and penalties etc. where applicable, as per contractual provisions.
- V. Regulation of payments to various agencies towards procurement of various project components and services in accordance with the contractual terms and conditions.
- VI. Regulation of payments to agencies engaged for capital and renovation works (Construction/Renovation works/other civil works carried out in office premises in UIDAI Hqrs/Regional Office etc).
- VII. While commenting on ROs, comment on Bank Reconciliation Statement and working of Tally may be made.

VIII. While conducting audit the billing done by/ payments made to Enrolment Agencies/Registrars (along with that of revenue earned by EAs) may also be audited and commented upon.

1.2 Non-financial issues

- I. Evaluation of the procedure and process for procurement of goods and services to be in accordance with the GFRs and the Procurement Manual of UIDAI.
- II. Review of tender documents/reports of TEC/Negotiation Committees constituted to ascertain whether due process as per the Procurement Manual was followed and they were approved by the Competent Authority as per the delegation of powers.
- III. Review of recording and maintenance of stores, inventory and consumables records as per rules. The review of inventory includes reviewing whether Physical verification was carried out as per rules.
- IV. Review of the procedure adopted for declaring stores/inventory a surplus unserviceable, disposal of scrap etc.
- V. Review of maintenance and oversight of other control records/registers (such as BG register, Cash book, PBR, Service books, Advance Register, Bill Register etc) rules as per rules.

2 Economy, Efficiency and Effectiveness issues:

This domain comprises review and analysis of various sub systems/project tracks in the office with a view to forming an opinion on the efficiency and effectiveness of the various processes and procedures towards achievement of the envisaged goals and objectives of the respective sub system/project track. The illustrative list of issues for review and analysis in this domain, including evaluation of the efficiency and effectiveness of underlying internal controls are as follows:

- I. Identification of the policy framework (where applicable) governing each project track/subsystem (Technology, Logistics, Authentication, Updation Facilitating development of Aadhaar Enabled Applications, ICT Assistance, facilitating deployment of Micro ATMs, DMS, KM Portal, UBCC, IEC, Training and testing, Content development, Accounting system, Payment processing system, etc) and assess its adequacy.
- II. Assessment of establishment goals and objectives vis-à-vis the policy framework (where applicable) in respect of the project track/subsystem bring reviewed.

- III. Assessment of the roles and responsibilities assigned to HQrs and to the Regional Offices in respect of each project track/sub system.
- IV. Assessment of the processes and procedures employed and assess the progress in respect of each project track/sub system with reference to roles/responsibilities assigned to the office being audited and achievement of the envisaged objectives/goals.
- V. Assessment of the utilization of resources (manpower, other infrastructure) with reference to the progress achieved so far in respect of each project track/sub system.
- VI. Identification and assessment of factors/risks inhibiting performance or goal fulfilment.
- VII. Suggesting improvements/mid-course corrections that may be necessary.

VIII. Review of the payment processing and accounting subsystems for efficiency and effectiveness would involve, among others, the process time lag from the date of receipt of invoices to the time it is passed for payment and accounted for under the relevant Heads of Expenditure.

3. UIDAI Income related Audit:

UIDAI earns income from Authentication Services, License Fee, Aadhaar Sewa Kendra Operations, Order Aadhaar Card Service and Self Service Update Portal. The Audit Teams conducting the Audit will also audit the billing done to AUAs/KUAs/ASAs. Revenue realized as well as any other income accrued to UIDAI and include in their report.

5.4.2 Ensuring GST Compliances of UIDAI

- I. Examining/vetting filed GST related returns of UIDAI and advising necessary corrective course measures.
- II. Framing best industry practices guidelines towards GST related Laws in UIDAI.
- III. Annual GST reconciliations and Audits.
- IV. Preparation and submission of replies against notices issued by the GST Departments in respect of UIDAI HO, ROs & Tech Centre.

Note: Bidder is expected to complete any other related work that is not explicitly detailed in the contract or agreement, in addition to the work that is specified above on the directions of UIDAI.

5.5.1 Timely performance of tasks

- a. In case all tasks as per the scope of the RFP are not completed in schedule as defined as part of SLAs, additional resources shall be provisioned by the successful Bidder without any additional cost.
- b. UIDAI may ask for additional number of resources as per rates mentioned in Commercial Bid in case of any increase of work or scope is there. It may also decrease the number of resources as per the requirements and the monthly payments will be adjusted accordingly.
- c. In case of attrition of any of the resources, Bidder will provide replacement and knowledge transfer at their own cost. Successful Bidder shall submit quarterly resource plan 15 days before commencement of each quarter for a quarter and to be approved by UIDAI. SLA performance relating to availability of resources shall be evaluated viz-a-viz quarterly resource plan.
- d. Attendance of resources will be considered as per UIDAI guidelines.

5.5.2 Removal and/or Replacement of Personnel

- a) If, for any reason beyond the reasonable control of the Bidder, any separation of resource from the Bidding firm, retirement, death, medical incapacity, among others, where it becomes necessary to replace any of the Personnel, the Bidder shall provide a replacement of the resource of equivalent or better qualifications.
- b) If the Purchaser:
 - (i) Finds that any of the Personnel have committed serious misconduct or have been charged with having committed acriminal action, or
 - (ii) Have reasonable cause to be dissatisfied with the performance of any of the Personnel, then the Bidder shall, at the Purchaser's written request specifying the grounds thereof, provide as a replacement a person with qualifications and experience acceptable to the Purchaser.
 - (iii) Any of the Personnel provided as a replacement under Clauses (a) and (b) above the Bidder shall bear all additional travel and other costs arising out of or incidental to any removal and/or replacement.

In the event of any request from the Purchaser to replace a resource on reasonable ground of non-performance or otherwise, the successful Bidder would be required to deploy a replacement of the resource within 30 days from receipt of written/email request from UIDAI. Bidder shall also ensure that in case of the replacement of any Personnel, knowledge transfer and its documentation is undertaken. There shall be 15 days of overlapping period for such knowledge transfer activity.

5.5.3 Service Levels Agreement

- 1. Finance Division is the one of the core Department of this organization. Hence, it shall be the bidder's responsibility to ensure availability of required resources during the entire duration of project. All tasks and timelines will be agreed as per UIDAI requirements and agreed upon within 1 month of signing of the Contract.
- 2. If the Vendor fails to meet the Service Levels within the Target time limit(s) as specified, the Purchaser shall without prejudice to its other remedies under the Contract, deduct the amount from the Contract value (liquidated damages).
- 3. In general, the number of hours working in day shall be 8 hrs 30 minutes and number of working days in a week shall be 5 days.
- 4. The resources can avail holidays as per UIDAI's list of holidays.
- 5. As and when necessary, based on criticality, it is expected that resources deployed will work extra hours including weekdays or weekend like financial year closure etc. without any additional payment.
- 6. The resources may avail up to a maximum of 4 days of leave in one quarter subject to cap of 12 days in a period of 12 months starting from the date of commencement of Services. However, the resources shall take prior permission from Purchaser for availing any leave.
- 7. For leaves days beyond the entitlements, deduction for absence per day will be calculated as "Monthly Manpower Cost/ 26"
- 8. The unused leaves at the end of the contract shall expire and there shall be no payment against the leave remaining unused.

- 9. The travelling and hotel expenses of resources deployed under this Contract for official tours carried out on the instruction of the Purchaser shall be borne by the Purchaser as per travel entitlements mentioned at Appendix E of this RFP.
 - 10. The overall liquidated damages will be maximum of 10% of the Contract Value.
- 11. SLA's that is applicable during the Contract period are as under:
- **A.** Delay in deployment of all agreed Manpower
 - a) Within 15 days of commencement of Contract No Penalty
 - b) For every day after the 15 Days Rs. 1000 per resource per day
- **B.** Submission of Draft Internal Audits report to Team leader
 - a) Within 2 Weeks of completion of Internal Audit No penalty
 - b) For every week after 2 weeks Rs 5000 per week per draft report
- C. Delay in completion of GST Reconciliation/Audits as per agreed schedule
 - a) Within agreed schedule No penalty
 - b) For every day after the agreed schedule Rs 500 per delay per day

D. Penalty for unauthorised absence beyond permissible leave days

S. No.	Designation	Penalty	
1.	Project Manager	Rs. 1,500/-	
2.	Deputy Manager	Rs. 1,000/-	
3.	Senior Audit Executive	Rs. 500/-	

E. UIDAI may consider termination of contract in case absence of Project Manager for more than 30 days.

6 SECTION VI - Appendix

6.1 Appendix A- Contract

CONTRACT

	<u>u</u>	<u>ONTIMICI</u>	
THIS AGI	REEMENT made on this	day of	between CEO, Unique
Identifica	tion Authority of India (hereinafter re	ferred to as "the Purchaser"),	acting through,
(Designat	cion), which expression shall unless re	pugnant to the context or mean	ning thereof mean and be
deemed t	o include its authorized representative	s and assigns of the FIRST PAR	Т;
		And	
	having its Office at	acting through	(designation)
(hereinaf	ter referred to as " the Vendor ")which	expression shall unless exclud	ed by or repugnant to the
context, ii	ncludes their Heirs, Executors, Adminis	strators and Legal Representati	ves of the SECOND PART.
WHEREA	S Purchaser is desirous of entering in	nto a contract for	with the
Vendor, a	and has accepted to pay to the Vendor t	the contract amount for provisi	oning of those Services at
a total val	lue not exceeding (Rupees) (I	Hereinafter referred to as "the G	Contract Value").
AND WH	EREAS the Vendor has agreed to provid	de the Services as listed in BidD	Oocument No, as per
the rate(s	s) given in the table below mentioned	hereinafter.	
NOW TH	IS AGREEMENT WITNESSETH AS FOL	LOWS:	
1. In	this Agreement words and expressi	ons shall have the same mea	nings as are respectively
assign	ned to them in the Conditions of Contra	ct referred to.	
	ne following documents (of Bid Docume rued as partof this Agreement viz:	ent/ RFP) shall be deemed to fo	orm and be read and
(i)	Invitation to Bid		Section I
(ii)	Instructions to Bidders		Section II
(iii)	General Conditions of Contract		Section III
(iv)	Contents of the bid		Section IV
(v)	Scope of Work		Section V
(vi)	Appendix		Section VI

	A. Bid Document Noregarding_	i	ncluding
	B. Clarifications issued by the Purchaser and su	ccessful Bidder.	
	C. Pre-Qualification, Technical and Commercial	Proposals submitt	ed by the Vendor.
	D. Order Nodatedpl	laced on the Vendo	r.
	E. Acceptance of the order vide No.	dated	by the Vendor.
3.	In consideration of the payments to be made mentioned, the Vendor hereby covenants with the therein in conformity in all aspects with the prov	ne Purchaser to pro	ovide the services and to remedy
4.	The Purchaser hereby covenants to pay the Vene as listed in Table below and the remedying of de as may become payable under the provisions prescribed by the Contract.	efects therein , the	Contract Value or such other sum
5.	TOTAL CONTRACT VALUE: Rs(Rupee	es)
the	eir respective laws the day and year first above wr Signed, sealed and delivered for &on behalf of M/s	Signed, sealed behalf of the C	and delivered for and on EO, Unique Identification
		Authority of In Identification A	ndia Authority of India
	Signature	_ Identification A	ndia Authority of India
	C: .	Identification A SignatureName	Authority of India
	Signature	Identification A SignatureName	dia Authority of India
	SignatureName	- Identification A Signature Name Designation	Authority of India
	Signature Name Designation	Name Designation Address	Authority of India
	Signature Name Designation Address	Name Designation Address	Authority of India
	Signature Name Designation Address Date	Name Designation Address Date	New Delhi
	Signature Name Designation Address Date Place : New Delhi	Name Designation Address Date Place	New Delhi
	Signature Name Designation Address Date Place : New Delhi In the presence of:	Name Designation Address Date Place In the presence	New Delhi
	Signature Name Designation Address Date Place : New Delhi In the presence of: Signature	Name Designation Address Date Place In the presence Signature	New Delhi
	Signature Name Designation Address Date Place : New Delhi In the presence of: Signature Name	Name Designation Address Date Place In the presence Signature Name	New Delhi
	Signature Name Designation Address Date Place : New Delhi In the presence of: Signature Name Designation	Name Designation Address Date Place In the presence Signature Name Designation	New Delhi

6.2 Appendix B - Performance Bank Guarantee

PROFORMA FOR CONTRACT PERFORMANCEBANK GUARANTEE BOND

Re	f: Date
Ba	nk Guarantee No
То	Unique Identification Authority of India Bangla Sahib Road, Behind Kali Mandir, Gole Market, New Delhi - 110001.
De	ar Sir,
1	consideration of the Unique Identification Authority of India, Ministry of Electronics & Information Technology, Government of India (hereinafter referred to as the
	"Purchaser" which expression shall unless repugnant to the context or meaning thereof include its successors, administrators and assigns) having awarded to M/s with its office at (hereinafter referred to as the "Contractor"/ "Vendor" which expression shall unless repugnant to the context or meaning thereof, include its
	successors, administrators, executors and permitted assigns), a Contract by issue of Notification of Award No
	Internal Audit functions in UIDAI" and the Contractor having agreed to provide a Performance Bank Guarantee for the faithful performance of the entire Contract not exceeding Rs(in words & figures).
2. V	We(Name & Address of Bank Branch) having its Head office at(hereinafter referred to as the "Bank", which expression shall, unless repugnant to the context or meaning

thereof, include its successors, administrators, executors and assigns) do hereby guarantee and undertake to pay the amounts due and payable under this guarantee without any demur, reservation, context, recourse or protest and/or without any reference to the Contractor merely on a demand from the Owner stating that the amount claimed is due by way of loss or damage caused to or would be caused to or suffered by the Owner by reason of breach by the said Contractor(s) of any of the terms or conditions contained in the said Agreement or by reason of the Contractor(s) failure to perform the said Agreement. Any such demand made on the Bank shall be conclusive and binding not withstanding any difference between the Owner and the Contractor or any dispute pending before any Court, Tribunal, Arbitrator or any other authority. We agree that the guarantee herein contained shall be irrevocable and shall continue to be enforceable till the Purchaser discharges this guarantee.

- 3. The Purchaser shall have the fullest liberty without affecting in any way the liability of the Bank under this guarantee, from time to time to extent the time for performance of the Contract by the Contractor. The Purchaser shall have the fullest liberty, without affecting this guarantee, to postpone from time to time the exercise of any powers vested in them or of any right which they might have against the Contractor, and to exercise the same at any time in any manner, and either to enforce or to forbear to enforce any covenants, contained or implied, in the Contract between the Purchaser and the Contractor or any other course or remedy or security available to the Purchaser. The Bank shall not be released of its obligations under these presents by any exercise by the Purchaser of its liberty with reference to the matters aforesaid or any of them or by reason of any other act of omission or commission on the part of the Purchaser or any other indulgences shown by the Purchaser or by any other matter or thing whatsoever which under law would, but for this provision have the effect of relieving the Bank.
- 4. The Bank also agrees that the Purchaser at its option shall be entitled to enforce this Guarantee against the Bank as a principal debtor, in the first instance without proceeding against the Contractor and notwithstanding any security or other guarantee the Purchaser may have in relation to the Contractor's liabilities.
- 5. This guarantee will not be discharged due to the change in the constitution of the Bank or the Contractor(s)/Service Provider(s)/ Vendor(s).

2025

RFP FOR ENGAGEMENT OF A CA FIRM/COMPANY FOR HANDLING INTERNAL AUDIT FUNCTIONS OF UIDAI

6. NOTWI	ithstanding anything contained hereinab	ove:
a.	Our liability under this Guarantee is res	tricted to (INR).
b. с.	time for such period(s) (not exceeding whose behalf this guarantee has been given whose liable to pay the guarantee amount of the such that th	
In witnes	s whereof the Bank, through its authoriz	ed officer, has set its hand and stamp on
this	day ofatat	
(Signat	 cure)	(Signature)
(Name	e)	(Name)
(Official A	Address)	(Designation with Bank Stamp)
		Attorney as per Power of Attorney No Dated

6.3 Appendix C - Bid Securing Declaration Form

"Bid Securing Declaration Form"

To

Unique Identification Authority of India

M/o Electronics & Information Technology, Government of India Bangla Sahib Road, Behind Kali Mandir, Gole Market, New Delhi – 110001

We, the undersigned, declare that:

We, M/s.....(herein referred as vendor) understand that, according to bid clause No. 2.6, bids may be supported with a Bid Securing Declaration

We will automatically be suspended from being eligible for bidding in any tender, request for proposal or other similar invite with the Unique Identification Authority of India (herein referred as Purchaser) for the period of three years, starting on bid submission closing date, if we are in breach of any of the following obligation(s) under the bid conditions:-

THE CONDITIONS of this obligation are:

- i. If a Bidder withdraws its bid during the period of bid validity specified by the Bidder in the Bid; or
- ii. If the Formats to prepare the price bid are found to be modified by the Bidder; or
- iii. In the case of a successful Bidder, if the Bidder fails:
 - to accept the notification of award and sign the contract in accordance with Section 2.16 of the RFP;
 - to furnish Performance Bank Guarantee in accordance with Section 2.17 Performance Bank Guarantee of the RFP.
 - If bidder withdraws its proposal during negotiations (failure to arrive at consensus by both the parties shall not be construed as withdrawal of proposal by the bidder).

2025

RFP FOR ENGAGEMENT OF A CA FIRM/COMPANY FOR HANDLING INTERNAL AUDIT FUNCTIONS OF UIDAI

We understand that this declaration shall expire if we are not the successful Bidder and on receipt of purchaser's notification of the award to another Bidder; or forty five days after the validity of the Bid; whichever is later.

(Signature)		
Authorized Signa	atory	
Name :		Designation:
Office Seal:		
Place:		
Date:		

6.4 Appendix D - Non-Disclosure Agreement

(To be provided on Non-judicial stamp paper of Rs.100/-)

WHEREAS,	we	the	unde	rsigned	Bi	idder,
			_, having our	r principal p	lace of busi	iness/
registered office a	nt		, are	desirous of	bidding fo	or Bid
No covering	5 " <u> </u>		_" (hereinaft	er called th	e said 'RF	P') to
the CEO, Unique	Identification Author	rity of India	a, having its o	office at Ban	ıgla Sahib F	Raod ,
Behind Kali Man	dir , Gole Market,	New Delhi	i - 110001,	hereinafter	referred	to as
'Purchaser' and,	WHEREAS, the Bio	lder is aw	are and con	firms that	the Purch	aser's
business/ opera	tions, information,	Applicatio	n/software,	hardware,	business	data,
architecture schei	matics, designs, stora	age media a	and other info	ormation / c	documents	made
available by the	Purchaser in the	RFP docur	nents during	g the biddi	ng process	s and
thereafter, or oth	nerwise (confidentia	ıl informati	ion for short	t) is privile	ged and st	trictly
confidential and/o	or proprietary to the	Purchaser,				
NOW THEREFORE	₹,					

In consideration of disclosure of confidential information, and in order to ensure the Purchaser's grant to the Bidder of specific access to Purchaser's confidential information, property, information systems, network, databases and other data, the Bidder agrees to all of the following conditions.

It is hereby agreed as under:

1. The confidential information to be disclosed by the Purchaser under this Declaration ("Confidential Information") shall include without limitation, any and all information in written, representational, electronic, verbal or other form relating directly or indirectly to processes, methodologies, algorithms, risk matrices, thresholds, parameters, reports,

deliverables, work products, specifications, architecture, project information, money laundering typologies, related computer programs, systems, trend analysis, risk plans, strategies and information communicated or obtained through meetings, documents, correspondence or inspection of tangible items, facilities or inspection at any site to which access is permitted by the Purchaser. We also hereby agree that this NDA will be binding on us through-out the contract period and will survive the contract period in case we are selected as a successful bidder.

- 2. Confidential Information does not include information which:
- a. the Bidder knew or had in its possession, prior to disclosure, without limitation on its confidentiality;
- b. information in the public domain as a matter of law;
- c. is obtained by the Bidder from a third party without any obligation of confidentiality;
- d. the Bidder is required to disclose by order of a competent court or regulatory authority;
- e. is released from confidentiality with the written consent of the Purchaser.

The Bidder shall have the burden of proving hereinabove are applicable to the information in the possession of the Bidder.

- 3. The Bidder agrees to hold in trust any Confidential Information received by the Bidder, as part of the Tendering process or otherwise, and the Bidder shall maintain strict confidentiality in respect of such Confidential Information, and in no event a degree of confidentiality less than the Bidder uses to protect its own confidential and proprietary information. The Bidder also agrees:
 - a. to maintain and use the Confidential Information only for the purposes of bidding for this RFP and thereafter only as expressly permitted herein;
 - b. to only make copies as specifically authorized by the prior written consent of the Purchaser and with the same confidential or proprietary notices as may be printed or displayed on the original;

- c. to restrict access and disclosure of Confidential Information to their employees, agents, and representatives strictly on a "need to know" basis, to maintain confidentiality of the Confidential Information disclosed to them in accordance with this clause; and
- d. to treat Confidential Information as confidential unless and until Purchaser expressly notifies the Bidder of release of its obligations in relation to the said Confidential Information.
- 4. Notwithstanding the foregoing, the Bidder acknowledges that the nature of activities to be performed as part of the Tendering process or thereafter may require the Bidder's personnel to be present on premises of the Purchaser or may require the Bidder's personnel to have access to software, hardware, computer networks, databases, documents and storage media of the Purchaser while on or off premises of the Purchaser. It is understood that it would be impractical for the Purchaser to monitor all information made available to the Bidder's personnel under such circumstances and to provide notice to the Bidder of the confidentiality of all such information.

Therefore, the Bidder shall disclose or allow access to the Confidential Information only to those personnel of the Bidder who need to know it for the proper performance of their duties in relation to this project, and then only to the extent reasonably necessary. The Bidder will take appropriate steps to ensure that all personnel to whom access to the Confidential Information is given are aware of the Bidder's confidentiality obligation. Further, the Bidder shall ensure that all personnel of the Bidder are bound by confidentiality obligation in relation to all proprietary and Confidential Information received by them which is no less onerous than the confidentiality obligation under this agreement.

- 5. The Bidder shall establish & maintain appropriate security measures to provide forthe safe custody of the Confidential Information & to prevent unauthorized access to it.
- 6. Confidential Information shall at all times remain the sole and exclusive property of the Purchaser. Upon completion of the Tendering process and/or termination of the contract or at any time during its currency, at the request of the Purchaser, the Bidder

shall promptly deliver to the Purchaser the Confidential Information and copies thereof in its possession or under its direct or indirect control, and shall destroy all memoranda, notes and other writings prepared by the Bidder or its Affiliates or directors, officers, employees or advisors based on the Confidential Information within a period of sixty days from the date of receipt of notice, or destroyed, if incapable of return. The destruction shall be witnessed and so recorded, in writing, by an authorized representative of the Purchaser. Without prejudice to the above the Bidder shall promptly certify to the Purchaser, due and complete destruction and return. Nothing contained herein shall in any manner impair rights of the Purchaser in respect of the Confidential Information.

7. In the event that the Bidder hereto becomes legally compelled to disclose any Confidential Information, the Bidder shall give sufficient notice and render best effort assistance to the Purchaser to enable the Purchaser to prevent or minimize to the extent possible, such disclosure. Bidder shall not disclose to a third party any Confidential Information or the contents of this RFP without the prior written consent of the Purchaser. The obligations of this Clause shall be satisfied by handling Confidential Information with the same degree of care, which the Bidder applies to its own similar Confidential Information but in no event less than reasonable care.

For and on behalf of:

(BIDDER)
Authorized Signatory
Office Seal:
Name & Designation:

Place: Date:

6.5 Appendix E - Frequency of Internal Audits and Travel Entitlements

6.5.1 Frequency of conducting Internal Audits

Sr. No.	Office Name	Required Frequency of Audits	Tentative Man days required	
1	H 1000 D III	0	20 days (5 days each * 4	
1	Head Office – Delhi	Quarterly		
			40 days	
2	8 Regional Offices	Annually	(5 days each * 8 Regional Offices)	
			10 days	
3	2 Tech Centers	Annually	(5 days each * 2 Tech Centers)	
			45 days	
4	Performance Audits <i>viz-a-viz</i> contractual payments to vendors of Functional Wings (15 divisions)	Annually	(3 days each * 15 Functional Wings)	
	Revenue Audit of Authentication		16 days	
5	Division	Quarterly	(4 days each * 4 quarter)	
			18 days	
6	6 State Offices	as and when directed	(3 days each * 6 State Offices)	
7	Other special Audits	as and when directed	5 - 10 days each	

6.5.2 Travel Entitlements Policy

- All business-related travels should be got approved by the Deputy Director General Finance.
 Only those travel expenses shall be passed that would have prior approval of the DDG Finance.
- The mode of Air travel for all deployed resources is mandated as Economy Class.
- The reimbursable expense limits on Lodging and Daily Allowance, applicable during the business trip and corresponding to each Location category, are set out in a Level wise entitlement matrix provided below: -

	'X' Class citi	es	'Y' Class citi	es	Remar	ks.
Resource position	Lodging	Food	Lodging	Food	i.	Cities Classifications: As per GOI. Order
Project Manager	5000	1000	4000	900		Number: 2/5/2017-E. II(B), dated 07
Deputy Manager	5000	1000	4000	900	ii.	July 2017.
Senior Audit Executive	4500	700	3500	600] 11.	All expenses are inclusive of applicable taxes.

- For Local travel for during Audit at locations other than Delhi all levels are entitled to Taxi.
- For any journey between locations that can be covered in 6 hours or less by train, the
 default mode of travel for all employees is AC Chair Car or II Tier AC as the case may
 be. Travel by Air will be based on the business exigency and with approval of DDG –
 Finance.
- To claim reimbursement, the agency needs to submit the Travel expense claim form to UIDAI along with the relevant bills/voucher, boarding passes, tickets and hotel bills and approval of the DDG – Finance.
- UIDAI reserves the right to review the Travel policy at its sole discretion.